

International Journal of Environmental Research and Public Health



# Article An Effective Approach for the Multiobjective Regional Low-Carbon Location-Routing Problem

# Longlong Leng, Yanwei Zhao \*, Jingling Zhang and Chunmiao Zhang

Key Laboratory of Special Equipment Manufacturing and Advanced Processing Technology, Ministry of Education, Zhejiang University of Technology, Hangzhou 310023, China; cyxlll@zjut.edu.cn (L.L.); jlzhang@zjut.edu.cn (J.Z.); zhcm@zjut.edu.cn (C.Z.) \* Correspondence: ywz@zjut.edu.cn

Received: 11 April 2019; Accepted: 10 June 2019; Published: 11 June 2019

Abstract: In this paper, we consider a variant of the location-routing problem (LRP), namely the the multiobjective regional low-carbon LRP (MORLCLRP). The MORLCLRP seeks to minimize service duration, client waiting time, and total costs, which includes carbon emission costs and total depot, vehicle, and travelling costs with respect to fuel consumption, and considers three practical constraints: simultaneous pickup and delivery, heterogeneous fleet, and hard time windows. We formulated a multiobjective mixed integer programming formulations for the problem under study. Due to the complexity of the proposed problem, a general framework, named the multiobjective hyper-heuristic approach (MOHH), was applied for obtaining Pareto-optimal solutions. Aiming at improving the performance of the proposed approach, four selection strategies and three acceptance criteria were developed as the high-level heuristic (HLH), and three multiobjective evolutionary algorithms (MOEAs) were designed as the low-level heuristics (LLHs). The performance of the proposed approach was tested for a set of different instances and comparative analyses were also conducted against eight domain-tailored MOEAs. The results showed that the proposed algorithm produced a high-quality Pareto set for most instances. Additionally, extensive analyses were also carried out to empirically assess the effects of domain-specific parameters (i.e., fleet composition, client and depot distribution, and zones area) on key performance indicators (i.e., hypervolume, inverted generated distance, and ratio of nondominated individuals). Several management insights are provided by analyzing the Pareto solutions.

**Keywords:** regional low-carbon location-routing problem; multiobjective optimization; fuel consumption; carbon emission; multiobjective hyperheuristics

# 1. Introduction

City logistics poses challenges to governments, businesses, carries, and citizens, particularly in the context of freight transport [1] in terms of three effects: economy, society, and the environment. Therefore, it requires new business operating models for addressing the above triple effects. Meanwhile, with increasing public demand for sustainable development and health, city logistics, as a primary source of carbon emissions (CE), should initiate reductions in CE during related activities. In the last few decades, many researchers have developed tools to optimize logistics networks, especially the location-routing problem (LRP), which includes location-allocation and vehicle-routing problems (VRP) [2]. However, the basic LRP model is only concerned with logistics costs and neglects the environmental and social effects of transportation.

Recently, the permutation and coordination of economic, social, and environmental benefits have emerged as one of the most addressed problem. The low-carbon LRP (LCLRP) concept introduced by Zhang et al. [3] focuses on depot locations and vehicle routes to minimize total CE

from depots and traveling activities. The regional LCLRP (RLCLRP), which was first proposed by Koc et al. [1], considers the distribution of clients and depots located in nested zones characterized by different speed limits. Leng et al. [2] further developed a single-objective model (SOM) for the RLCLRP, with consideration of a set of given real-world constraints. The above three papers all share the same feature, namely SOMs. In real-life, however, multiobjective problems (MOP) always play a role in determining trade-offs among multiple demands. Therefore, Leng et al. [4] developed a biobjective model for the RLCLRP and quantum-inspired selection to improve hyper-heuristic

performance. Meanwhile, for the other variants of LRP, there were several studies devoted to the development of economic and environmental effects. For example, Pourhejazy et al. [5] considered the fuel consumption in a variant of LRP, and modeled a biobjective model for their problem. With the development of economy and society, joint effects of logistics on economy, society, and

environment should be considered simultaneously rather than separately. As the main performance indicators of logistics network, our proposed model is in tune with the above triple concerns. In other words, the first objective defined FCCE as a traveling cost (economic and environmental effect), delivery duration (i.e., total traveling time) as the second objective, and client satisfaction (social effect) (i.e., total client waiting time) as the third objective.

In real world, most logistics enterprises don't consider and analyze the joint effects in city logistics as they are only pursuing economic profit. Additionally, our study follows the basic idea of Leng et al. [4] and defines a multiobjective model (MOM) for the RLCLRP (MORLCLRP). To date, no research has been conducted using MOMs as a means to minimize logistics cost, service duration, and client waiting time, simultaneously. Furthermore, fewer studies have incorporated fuel consumption and carbon emission (FCCE) into basic LRP models, such as Pourhejazy et al. [5] and see Section 2.3. Indeed, most studies have failed to handle FCCE appropriately, in other words, the FCCE should be used as the traveling cost like [1–3] instead of an objective or constraints (see Section 2.2). The main contributions are as follows:

- 1. We introduced a mixed-integer linear programming MOM for the RLCLRP. In the real-world logistics network, cost, service duration, and client satisfaction were considered the most significant performance indicators, and traveling cost incorporated FCCE cost.
- We developed an efficient MOHH to solve the MORLCLRP. For HLHs, we provided four selection strategies and three acceptance criteria to improve the performance of the MOHH framework and developed three MOEAs as the pool of LLHs.
- 3. We conducted extensive computational experiments to assess the efficiency of the proposed algorithms and developed managerial implications by assessing problem parameters, such as client and depot locations, speed zone areas, and fleet composition. The model, algorithms and computational results can serve as a stepping-stone for further MORLCLRP research with cold chain logistics [6].

The paper is structured as follows: Section 2 provides a review of related literature; Section 3 describes the MORLCLRP with simultaneous pickup and delivery, heterogeneous fleet, and hard time windows; Section 4 gives a brief description of the hyper-heuristic framework together with a general MOEA structure for the MORLCLRP; Section 5 describes the computational experiments and simulated results; and, Section 6 outlines the study conclusions.

## 2. Literature Review

An important objective of the current research was to acquire a set of Pareto solutions for the MORLCLRP within the logistics network. We reviewed related literature based on models estimating FCCE, factors affecting FCCE, and the LCLRP and RLCLRP. As to the literature about the solution methods for the LRP and its variants, the reader is referred to the surveys of Pourhejazy et al. [7] and Drexl and Schneider [8].

A good estimation model requires analysis of influencing factors. Ardekani et al. [9], Bigazzi and Bertini [10], Alwakiel [11], Lin et al. [12] and Demir et al. [13,14] investigated the factors influencing FCCE, with five categories being identified as a result: (1) vehicle-specific, including rough mass, shape, engine size/type, and fuel type; (2) environmental, including roadway gradient, wind, ambient temperature, altitude, pavement type, and surface conditions; (3) traffic conditions, including congestion, speed limit, and traffic signal; (4) driver behavior, including operation level (e.g., shifting skill) and idle control; and (5) operational-related including fleet size and composition, load, and number of stops. Among the above factors, vehicle speed and vehicle-specific parameters are the most significant.

As speed is affected by inertia, air/rolling resistance, and road slope, it is one of the most important factors affecting FCCE [14]. Koc et al. [1] determined the optimal speed that minimizes FCCE. Demir et al. [15] also derived an optimal speed and demonstrated that reduction in FCCE can be achieved under varying speed environments, as also concluded by Poonthalir and Nadarajan [16]. Moreover, in existing FCCE models, the basic distinction between the two groups (i.e., micro and macro views, see Section 2.2) is the simulated model of vehicle speed (i.e., constant or non-constant). From the perspective of transportation activities, three factors influence vehicle speed, including traffic conditions, driver behaviors, and speed limits. Traffic conditions are characterized by traffic flow, especially during rush hour. This factor includes a step function to simulate varying speed, such as the functions proposed by Kuo [17], Kazemian and Aref [18], and Mirmohammadi et al. [19]. Driver behaviors is the main factor affecting gear selection, acceleration/deceleration, and idle control. Poonthalir and Nadarajan [15] used triangular distribution to simulate vehicle speed fluctuation between a minimum and maximum speed and stated that vehicles travel at several "most likely" speeds, indicating that a driver may select chronic gears at these speeds to complete transportation activities. Speed limits are an especially important factor on city roadways. Koc et al. [1] proposed speed zones to describe speed limitations according to real traffic roadways, with each zone having a speed limit and nested nature. Leng et al. [2,4] also applied speed zones, named "regional" representation.

Among the various factors affecting FCCE, fleet size and type are also important in most distribution activities [1,20–22]. Koc et al. [21] illustrated the benefits of a mixed fleet in reducing CE. Koc et al. [1] also demonstrated that a mixed fleet could reduce logistics costs and capacity utilization rates. Pitera et al. [23] explored rules-of-thumb for vehicle assignment within a mixed fleet to clarify simple implementations, such as assigning cleaner vehicles to routes with more clients and longer travel distances. Xiao et al. [24] emphasized that mixed fleets are concerned with individualized features including fleet types, CE rates/models, and load capacities.

## 2.2. Research about FCCE Models

Demir et al. [13] state that transportation activities can have damaging on environmental and public health by producing harmful emissions. Thus, accurate estimation models are required to measure and reduce FCCE during planning. To date, a variety of different analytical estimation models have been developed. According to Demir et al. [14] classified, three increasingly complex model types have been proposed, as described in Table 1:

Module	Characteristic	Example
Factor	Only consider a few key factors,	Fuel consumption rate (FCR) [3,6,25–30]; models by Pourhejazy et
model (FM)	such as travel distance, vehicle load.	al. [5]
Macro view	Average aggregate network	Computer programs to calculate emissions from road
	parameters, factors don't change	transportation [31,32]; Methodology for calculating transportation
(Macro)	over traveling time	emissions and energy consumption [33,34], etc.
Micro mou	More detailed information second	Comprehensive model emission model (CMEM) [1,2,4,13,15,21];
(Micro)	lovel econogially vehicle encod	comprehensive power-based fuel consumption models [35];
	level, especially vehicle speed.	vehicle specific power model [36.37]

Table 1. Classification of models estimating FCCE.

The above three model groups are categorized by complexity. Demir et al. [14] also discussed load (power)-based and regression-based emission models. The former investigates vehicle gross weight and vehicle-specific parameters, such as fuel type, engine, and size. The latter considers prediction models by statistically analyzing the relationship between FCCE amount and its factors. For example, the Japanese government [38] indicated that travel distance per volume unit of fuel used is strongly correlated to vehicle gross weight, which forms the basis of the FCR model. Among the above three model types, FM/Micro is simplified/complicated version of Macro, and the nature of conversion is available between FM/Micro and Macro. Actually, from an instantaneous estimation perspective, FM is a part of Macro, because the parameters in FM, namely vehicle-specific and travel speed parameters, are negligible or constant. Therefore, the model types can be classified into non-instantaneous and instantaneous models. From the perspective of accuracy for estimating FCCE, Micro views are the best, followed by Macro, and then FM. To better understand FCCE models, please see Lin et al. [12], Demir et al. [14], and Demir et al. [13].

## 2.3. Research Concerning LCLRP

Environmental and public health require the sustainable development of the environment, economy, and society. Low-carbon supply chain network design has become an important area of research [39]. As an essential tool in the supply chain, the LRP should consider reduction of FCCE. Various researchers have analyzed and modeled the LCLRP and its variants:

- Mohammadi et al. [40] proposed a biobjective LCLRP by optimizing logistics costs and traveling distance (i.e., low carbon/green objective). Although traveling distance is a significant factor in FCCE, other parameters, especially vehicle-specific parameters and load, should be considered in the model;
- Govindan et al. [41] presented a more detailed model by combining traveling and fixed CE of depots and manufacturers, with several coefficient matrices (e.g., uniform distribution) used to consider the effect of vehicle and load. However, these coefficient matrices are generated with uniform distribution, which does not accurately reflect FCCE; This approach was also applied in Chen et al. [39] and Nakhjirkan and Rafiei [42];
- Validi et al. [43–45] used fuel efficiency and distance to calculate the FCCE, like Kuo et al. [17] and Poonthalir and Nadarajan [16]. Faraji and Afshar-Nadjafi [46] proposed a modified method considering the extra fuel consumption caused by carrying each extra loads. Tang et al. [47] applied the method to calculate routing CE by giving parameters for each edge, with the CE of depots/inventory also considered;
- Qazvini et al. [48] proposed a SOM considering the cost of fuel consumption as a constraint. Although this method is effective, it is more appropriate to view the FCCE cost as a traveling cost in the real world;
- Koc et al. [1] considered a city in which goods need to be delivered from a depot to clients located in nested zones characterized by different speed limits and used CMEM to estimate FCCE, which was considered a traveling cost. This followed Leng et al. [2], who studied an extensive version of the RLCLRP using a shared mechanism-based hyperheuristic. Leng et al. [4] also modeled a biobjective RLCLRP tackled by quantum-inspired MOHH;
- Rabbani et al. [49] developed a MOM for the LCLRP with a heterogeneous fleet. Among the three objectives, cost, distance, and CE were viewed as contradictions. Fuel consumption rate, similar to Pourhejazy et al. [5], was used;
- Toro et al. [50] deduced the FCCE model by analyzing the forces acting on a vehicle and found the model to be remarkably similar to the CMEM. In their MOM, they also looked at fuel consumption and total emissions associated with the fuel consumption model;
- Wang and Li [51] applied the FCR considering road slope as a FCCE model. However, in their MOM, somewhat confusingly, both objectives were cost objectives, and penalty and vehicle fixed costs were added with FCCE cost;
- Zhang et al. [3] applied the FCR model to calculate CE and used the quantum evolutionary algorithm to solve the proposed model. Wang et al. [30] subsequently developed an ant-based

hyperheuristic to solve the model by Zhang et al. [3]. Leng et al. [28] proposed an extensive version of the model by Zhang et al. [3], and a quantum-inspired hyperheuristic to solve it. Zhao et al. [29] also developed an integrated model for the LCLRP by defining the FCCE cost, and developed an evolutionary hyperheuristic to solve it; Qian et al. [52] modified the model by Zhang et al. [3] with a biobjective model, and used tabu search-based MOHH to solve it;

• Wang et al. [27] developed a SOM using the FCR to estimate FCCE and as a part of costs.

From the above LCLRP-based papers, four methods were used for estimating FCCE: i.e., distance-based [39–42], fuel efficiency and distance-based [5,43–49], load and distance-based [3,27–30,52], and microscopic [1,2,4,50]. In addition, five modes of processing FCCE can be summarized: SOM using FCCE amount [3,28,30], SOM using FCCE as a cost [1,2,4,27,29,42], SOM using FCCE as a constraint [48], MOM using FCCE as an objective [39–41,43–47,49,50,52], and MOM using FCCE as a cost [4]. With continued development and attention, the LRP with consideration of environmental effects will become a focus of future operational research. However, in the real-life supply chain, FCCE cost should be added as a traveling cost, which also should include road tolls and parking fees, as per Koc et al. [1] and Leng et al. [2,4,29]. Therefore, we applied the same method [4] and defined FCCE as a traveling cost as the first objective (economic and environmental effect), delivery duration (i.e., total traveling time) as the second objective, and client satisfaction (social effect) (i.e., total client waiting time) as the third objective. For the solution approach, we reviewed recent literature regarding MOHH in the following section.

#### 2.4. Research about MOHH

Metaheuristics have been widely tailored for different domains. However, selection of the best algorithm and configuration of parameters and operators/pairs to solve problems can be difficult and time-consuming [53,54]. Moreover, even though a single tailored-search solution may achieve good performance, it is impossible to solve all cases with high efficiency. Therefore, the definition of hyperheuristic emerges while in such situations. Walker et al. [55] claimed that hyper-heuristic approaches are gaining renewed interest due to the development of more applicable search methods. The raw hyper-heuristic ideal was initially derived by Denzinger et al. [56], with the basic concept further developed by Cowling et al. [57] as "heuristics to select heuristics". Burke et al. [58] then proposed a comprehensive definition including (1) heuristic selection and (2) heuristic generation. The difference between heuristic selection and generation is recognized by the initial solutions; the second applies LLH to generate complete solutions beginning with empty solutions, whereas the first is initialized with complete solutions.

Metaheuristics are domain-specific, fine-tuned, and tailored-made rules-of-thumb that provide a set of mechanisms to search the domain space directly with the goal of obtaining optimal or nearoptimal solutions. However, the purpose of hyperheuristic is to improve the level of generality and portability by selecting/generating appropriate operators (i.e., control operator space) to optimize rather than directly modify the domain-specific space. In other words, compared with customized methods, hyper-heuristic approaches are generally applicable to other problem domains as they contain no domain knowledge. Although Kumari and Srinivas [59] demonstrated that domainspecific, fine-tuned, and tailored-made methods will unquestionably outperform a general framework, hyperheuristics sharing the same/similar operators would also obtain good, possibly better solutions [2,28–30,53].

The framework of the selection hyperheuristic involves HLH and LLH. The former searches the space formed by a set of LLHs, which directly optimize the space of domain-solutions [60]. Two decisions are defined in HLH: i.e., selection strategy and acceptance criterion. Selection strategy controls and monitors the performance of each LLH to intelligently choose the appropriate operators, with a single candidate solution then generated. After that, the new solution is either accepted or rejected. For LLH, all domain-specific knowledge is provided, including encoding, decoding, chromosome, and heuristics information, such as operators, crossovers, mutations, local searches, or metaheuristics [60]. It is worth noting that, a domain-barrier is developed to prevent domain knowledge from LLH to HLH; however, HLH is allowed to access problem domain-independent

information, such as number of operators, count/time of applied operators, fitness rate improvement, and fitness/objective values.

To put the selection strategy into perspective, three main types have been observed according to the source of feedback information: (i) online-learning, which occurs when a hyperheuristic is performed and includes choice function (CF) [2,4,28,57], multi-armed bandit [2,4,28,53–55], ant-based selection [29,30], and quantum-inspired selection (QS) [4,28,29]; (ii) offline-learning, which applies rules or programs to gather information by training a set of instances, including learning classifier systems, case-based reasoning, and genetic programming; and (iii) no-learning, which does not use any feedback information from search space, including simple random (SR), random descent, and random permutation. In contrast, there are two main types of acceptance criterion: (i) deterministic acceptance, which applies a 0–1 method-based strategy, including all moves (AM), improving and equal, and only improving; and (ii) nondeterministic acceptance, which calculates acceptance probability to judge the acceptance of new solutions, including simulated annealing, great deluge acceptance (GDA) [61], and late acceptance (LA) [61]. For a comprehensive understanding of hyperheuristics, please see Chakhlevitch and Cowling et al. [62], Burke et al. [58,63].

Although hyperheuristics have mainly been applied in SOMs, such as VRP [55], LRP and its variants [2,28–30], and sales summit [57], many studies have utilized hyperheuristics to tackle MOPs. For brief review of MOHH before 2014 see Maashi et al. [64], Table 2 provides a list of MOHH applications in MOPs since 2014, including the source of publication, published year, domain-problem, and main characteristics/components.

Authors [Reference] (Year)	Domain-Problem	Main Characteristics/Component
Koulinas et al. [65] (2014)	Resource constrained project scheduling problem	Domain-LLH; particle swarm optimization
Kateb et al. [66] (2014)	Runtime usage of MOEAs	Artificial selection of mutation for NSGAII
Maashi et al. [64] (2014)	MOEAs benchmark (WFG test suite)	NSGAII, SPEA2, MOGA; choice function; All moves
Castro and Pozo [67,68] (2014&2015)	MOEAs benchmark (DTLZ test suite)	Using leader selection methods and archiving strategies as LLH; IE acceptance; R2 indicator
Goncalves et al. [69] (2015)	MOEAs benchmark (UF test suite)	Using DE operators as LLH; MOEA/D; Tchebycheff function
Hitomi and Selva [70] (2015)	Architecture optimization problems	Dynamic MAB; using heuristic agents as LLH
Maashi et al. [61] (2015)	MOEAs benchmark (WFG test suite)	Choice function; great deluge algorithm and late acceptance; NSGAII, SPEA2, and MOGA
Qian et al. [71] (2016) Kumari and Srinivas	Multiobjective optimization	Using selection mechanisms, mutation, acceptance strategies as LLH Using crossover and mutation as LLH; NSGAII ranking mechanism;
[59] (2016) Strickler et al. [54]	variability test of feature	reinforcement learning strategy with adaptive weights Fitness rate rank-based MAB; NSGAII ranking mechanism; crossover and
(2016) Eroitag and	models Schoduling rules for complex	mutation operators
Hildebrandt [72] (2016)	manufacturing systems	Using scheduling rules as LLH; simulation-based genetic programming;
Guizzo et al. [73] (2017)	Integration and test order problem	Choice function, MAB; crossover and mutation operators; reward based- Pareto dominance
Li et al. [60] (2017)	Wind farm layout problem	Random, fixed sequence, choice function; all moves, GDA, NSGAII ranking mechanism
Ferreira et al. [53] (2017)	Software Product Line Testing	Using crossover and mutation operators as LLH; HH-based MOEAs (IBEA, SPEA2, NSGAII, and MOEA/D-DRA), namely elitism selection strategy ; random and upper confidence
Hitomi and Selva [74] (2017)	MOEAs benchmark (WFG, UF and DTLZ test suite)	HH-based MOEAs (IBEA, MOEA/D-DRA, NSGAII); multiple adaptive operator selections
Xu et al. [75] (2018)	Multiobjective mapping for network-on-chip	Genetic-based hyper-heuristic algorithm; using genetic operator as LLH; reinforcement learning strategy with adaptive weights
Yao et al. [76] (2018)	Multiobjective route planning in a smart city	Reinforcement learning mechanism; domain-LLH;
Almeida et al. [77] (2018)	Permutation flow shop problem	MOEA/D-DRA-based MOHH; MAB; crossover and mutation operators
Gomez and Terashima- Marin [78] (2018)	Bi-objective 2D bin packing problems	Evolutionary hyper-heuristics multiobjective framework; NSGAII, SPEA2 and GDE3
Castro et al. [79] (2018)	MOEAs benchmark (DTLZ, WFG test suite)	Fitness rate rank-based MAB; using leader selection methods and archiving strategies as LLH; R2-based IE acceptance and reward
Zhang et al. [80] (2018) Qian et al. [52] (2018)	Software release planning MOLCLRP	Extreme value credit assignment; probability matching; domain-LLH Tabu search; reinforcement learning method; NSGA-II ranking mechanism

Table 2. Applications of MOHH since 2014.

Zhou et al. [81] (2019)	Flexible job shop scheduling	NSGA-II ranking mechanism; Pareto strength; genetic programming
Chand at al [82] (2010)	resource constrained project	genetic programming hyperheuristic; NSGA-II ranking mechanism; using
Chand et al. [82] (2019)	scheduling problem	priority rules as LLH
Li et al. [83] (2019)	MOEAs benchmark	Learning automata; MOEAs as LLHs

Based on the development of MOHHs in Table 2, LLHs can be operators and metaheuristics, and thus two main modules can be obtained: (1) MOEA-based hyperheuristic (MOHH-I) which utilize operators (i.e., crossover, mutation, and domain-specific operators) or components (such as leader selection methods and archiving strategies in MOEAs [65,67-68,79]) as LLHs and NSGA-II corresponding elitism selection strategies, such as ranking mechanism [54,57,59,60,74,78,81,82] and Pareto Strength (SPEA2) [57,78], as acceptance criteria. Here, HLHs can access the objective value [60] and the methods involving objective space instead of solution space are independent of the domain problem. (2) Basic MOHH-II, which use MOEAs as LLHs, e.g., Li et al. [60,83] and Maashi et al. [61,64], and design corresponding acceptance criterion, such as AM [61,64], improving and equal [67,68,79], GDA and LA [64,83]. However, this classification is not shared by other studies.

MOHHs have been successfully developed in both continuous and combinational optimization problems ("domain-problem" column in Table 2). To the best of our knowledge, only our previous studies have applied MOHHs to tackle the MORLCLRP, that is, Qian et al. [52] and Leng et al. [4]. Regretful, no other researcher has yet applied MOHHs to solve MOMs of the LRP, or VRP. The reasons for this may be as follows: (i) compared with other domains, much more time and effort are needed for programing effective LLHs of the MORLCLRP; (ii) greater computing time is required to execute the MORLCLRP algorithms; and (iii) the MORLCLRP is a fairly complex NP-hard problem, which means that it can be difficult to achieve a good Pareto set. Therefore, corresponding operators and metaheuristics for the MORLCLRP are urgently needed.

#### 3. Mathematical Model

The applied model of estimating FCCE amount is the CMEM, and the corresponding parameters in the CMEM can be obtained from Ref. [1,2]. The descriptions and assumptions of domain problem are given in Section 3.1; Section 3.2 conducts the corresponding mathematical formulation and the necessary constraints; finally, we also provide some valid but unnecessary restrictions on the MORLCLRP in Section 3.3.

## 3.1. Description and Assumption of MORLCLRP

The MORLCLRP can be described on a complete and directed graph  $\Omega = (N, E)$  with a vertex set N and an edge set of E. N consists of a subset of D of  $N_d$  candidate depots and a subset of  $C = N \setminus D$  of  $N_c$  clients. Each client  $c \in C$  has a non-negative pickup demand  $p_c$  and delivery demand  $d_c$ , to be served exactly once, and is assigned to a single depot  $d \in D$  with capacity  $w_d$  and renting cost  $FD_d$ . The shipment of clients demand from the chosen depot is carried out by an unlimited set  $V = \{v_1, v_2, v_3\}$  of heterogeneous vehicles with capacity  $Q_v$  and one-time renting cost  $FV_v$ ; and traveling fuel consumption amount  $W_{ijv}$  of each edge  $(i, j) \in E = \{(i, j): i, j \in N, i \neq j\} \setminus \{(i, j): i, j \in D\}$  is calculated by the CMEM model, and it is worth noting that the distance of each edge  $(i, j) \in E$  are obtained by taxicab geometry (see Koc et al. [1] and Krause [84]).

The following are the hypotheses made: (1) each client is visited only once by a single vehicle and depot; (2) each vehicle must return the original depot from where it departs; (3) the load of each vehicle must be less than its vehicle capacity in service time; (4) the depot must serve the clients assigned to it; and (5) the vehicle must arrive at each node before the closing time windows.

#### 3.2. Mathematical Model

To formulate the problem, we define the following additional decision variables. Let  $x_{ijv}$  be equal to 1 if a vehicle of type  $v \in V$  travels on edge  $(i, j) \in E$  and to 0 otherwise; let  $y_i$  be equal to 1 if a depot  $j \in D$  is selected and to 0 otherwise; let  $z_{ij}$  is equal to 1 if client  $i \in C$  is serviced by depot  $j \in D$  and to 0 otherwise. A multiobjective formulation of MORLCLRP is given by:

$$\min f_1 = \sum_{j \in D} FD_j y_j + \sum_{i \in C} \sum_{j \in D} \sum_{v \in V} FV_v x_{ijv} + (f_{FC} + \zeta f_{CE}) \sum_{i \in N} \sum_{j \in N} \sum_{v \in V} W_{ijv} x_{ijv}$$
(1)

$$\min f_2 = \sum_{i \in C} \sum_{j \in D} \sum_{\nu \in V} AT_j x_{ij\nu}$$
(2)

min 
$$f_3 = \sum_{j \in N} \sum_{i \in C} \sum_{v \in V} (max\{l_i - AT_i - ST_i, 0\}) x_{jiv}$$
 (3)

where  $f_{FC}$  and  $f_{CE}$  are the unit price of 1 L of fuel and 1 kg of CE, respectively;  $AT_j$  is the time which service starts at node  $j \in N$ . Objective (1) is to minimize the total costs consisting of opened depots, vehicle, FC, and CE costs, corresponding to the economic and environmental effects, and the reason of simultaneously using FC and CE is that they represent the costs of consuming fuel and purchasing carbon emission from carbon trading market. Objective (2) aims at optimizing the total service time, corresponding to the long-term benefit. And the clients' satisfaction is modeled by the total client waiting time in objective (3), corresponding to the social interest. The constraints (4–23) are the necessary restrictions for the MORLCLRP.

The following two are degree restrictions, in particular constraint (4) makes sure that each client must be visited only once, and constraint (5) ensures that entering and leaving arcs to each node are equal:

$$\sum_{j \in N} \sum_{v \in V} x_{ijv} = 1, \ i \in C$$

$$\tag{4}$$

$$\sum_{j \in N} \sum_{v \in V} x_{jiv} = \sum_{j \in N} \sum_{v \in V} x_{ijv}, \ i \in C$$
(5)

Constraints (6 and 7) ensure that each client must be visited by a single depot and vehicle:

$$\sum_{j\in D} z_{ij} = 1, \ i \in \mathcal{C} \tag{6}$$

$$x_{ijv} + \sum_{k \in N} \sum_{p \in V, p \neq v} x_{jkp} \le 1, \ i \in N, j \in C, v \in V$$

$$\tag{7}$$

Restrictions (8–10) define illegal routes, i.e., each vehicle must return back to the original departure depot:

$$\sum_{v \in V} x_{ijv} \le z_{ij}, \ j \in D, i \in C$$
(8)

$$\sum_{v \in V} x_{jiv} \le z_{ij}, \ j \in D, i \in C$$
(9)

$$\sum_{v \in V} x_{ijv} + z_{ik} + \sum_{m \in D, m \neq k} z_{jm} \le 2, \ k \in D, i, j \in C$$

$$(10)$$

The above three restrictions have been proved by Karaoglan et al. [85]. Constraints (11) ensures that total demand supplied/picked up by a depot cannot exceed its capacity; constraints (12) and (13) are the extra bounds about the load when all vehicles depart from and return back to the depot:

$$max\left\{\sum_{i\in C} p_i z_{ij}, \sum_{i\in C} d_i z_{ij}\right\} \le w_j y_j, \ j \in D$$
(11)

$$\sum_{j \in C} \sum_{v \in V} L_{ijv} = \sum_{j \in C} d_j z_{ji}, i \in D$$
(12)

$$\sum_{j \in C} \sum_{\nu \in V} L_{ji\nu} = \sum_{j \in C} p_j z_{ji}, i \in D$$
(13)

where the  $L_{ijv}$  is the dynamic load of a vehicle of type  $v \in V$  traveling over an edge  $(i,j) \in E$ . The dynamic load of each vehicle must not exceed the its capacity, which is ensured by constraint (14); the relaxed (if  $d_i < p_i$ ) or intensified (if  $d_i > p_i$ ) is given in restriction (15); constraint (16) implies that the delivery and pickup demand of each client is satisfied; constraints (17) and (18) indicate that the load of vehicle

of type  $v \in V$  must be equal to the total delivery/pickup demand when departs from/return back to the depot:

$$0 \le L_{ijv} \le Q_v x_{ijv}, \ i, j \in N, v \in V \tag{14}$$

$$(d_j - p_j)x_{ij\nu} \le L_{ij\nu} \le (Q_\nu - d_j + p_j)x_{ij\nu}, \ i \in N, j \in C, \nu \in V$$
(15)

$$\sum_{i \in N} \sum_{v \in V} (L_{ijv} - d_j) x_{ijv} = \sum_{i \in N} \sum_{v \in V} (L_{jiv} + p_j) x_{ijv}, \ j \in C$$
(16)

$$\sum_{i\in D}\sum_{j\in C} L_{ij\nu} = \sum_{i\in C}\sum_{j\in N} d_i x_{ij\nu}, \ \nu \in V$$
(17)

$$\sum_{i\in C}\sum_{j\in D}L_{ij\nu} = \sum_{i\in C}\sum_{j\in N}p_i x_{ij\nu}, \ \nu \in V$$
(18)

The next two are used to restrain the vehicles' behavior according to clients' time windows:

$$AT_{j} = (max\{AT_{i}, e_{i}\} + ST_{i} + TT_{ij})x_{ijv}, \ i \in C, j \in N, v \in V$$
(19)

$$0 \le AT_i \le l_i, \ i \in C \tag{20}$$

where *ST*<sub>i</sub> corresponds to the service time of node  $i \in C$ ; *TT*<sub>ij</sub> is the travel time over an edge  $(i,j) \in E$ .  $(e_i, l_i)$  represents the time windows of client  $i \in C$ . Looking at constraint (19), if a vehicle arrives at client  $i \in C$  before time  $e_i$ , it waits until  $e_i$  to start to serve the client. Constraint (20) is designed for assumption (5). The next inequality is to restrict that opened depots must serve clients:

$$z_{ij} \le y_j, \ i \in C, j \in D \tag{21}$$

$$\sum_{i\in C} z_{ij} \ge y_j, \ j \in D$$
(22)

Constraint (21) ensures that an unselected depot mustn't service the clients; constraint (22) makes sure that an opened depot must serve clients. The above constraints are significant for the proposed problem, and we introduce some extra valid restrictions but not necessary in Section 3.3.

## 3.3. Other Valid Constraints

We introduce several polynomial size valid inequalities for our problem, which were used by several studies for VRPs/LRPs. The first inequality is the subtour elimination:

$$x_{ijv} + x_{jiv} \le 1, \ i, j \in C, v \in V$$
 (23)

Constraint (23) derived from Koc et al. [22] are only suitable for the directed graph with specific constraints, such as time windows and simultaneous pickup and delivery, instead of the basic LRP and heterogonous fleet. The next inequality is to restrict the opened depots and vehicles:

$$\sum_{j \in C} x_{ijv} \le y_i, \ i \in D, v \in V$$
(24)

$$\sum_{i \in C} \sum_{v \in V} x_{ijv} \ge y_j, \ j \in D$$
(25)

Constraint (22) ensures that an unselected depot mustn't service the clients; constraint (23) makes sure that an opened depot must serve client; constraint (24) guarantees that an unselected depot cannot assign vehicles; constraint (25) imposes that an opened depot must have vehicles to serve clients. The third restriction is introduced to the limitation of vehicles:

$$\left|\frac{\max\{\sum_{i\in C} d_i, \sum_{i\in C} p_i\}}{\max_{v\in V}(Q_v)}\right| \le \sum_{i\in D} \sum_{j\in C} \sum_{v\in V} x_{ijv} \le N_c,$$
(26)

Constraint (26) provides a lower and upper bound on the number of vehicles originating from depots, and [•] is the smallest integer larger than •. The fourth valid inequity is the number of opened depots:

$$\sum_{i \in D} w_i y_i \ge max \left\{ \sum_{i \in C} d_i , \sum_{i \in C} p_i \right\}$$
(27)

Constraint (27) states that the total capacity of opened depots is larger than the maximum demand. However, this restriction is not necessary, since sometimes if it is met, but it cannot provide restriction. The classical example is the Perl83-55×15 in Barreto et al. [86].

$$\sum_{j \in C} x_{ijv} + \sum_{j \in C} x_{jgv} \le 1, \ i, g \in D, i \neq g, v \in V$$

$$(28)$$

The final valid inequity is a complementary for constraint (10), which forbids the different depots in a single route. Looking at the extra valid restrictions, most of them can be expressed by the constraints (4–23). However, the aim of this section is to help readers easily understand the proposed problem.

## 4. Proposed Method

This section describes the proposed MOHH from two aspects: (1) domain-specific level and (2) high-level strategy. Section 4.1 introduces the necessary chromosome representation, a general framework for MOEAs as LLHs, and practical operators. Section 4.2 provides the MOHH framework for solving the proposed MORLCLRP. Section 4.3 discusses the four selection strategies. Section 4.4 discusses the three acceptance criteria.

#### 4.1. Domain Method

### 4.1.1. Chromosome Representation

In the proposed algorithm, the chromosome represents a complete solution, i.e., a collection of routes. Each route is stored in the cell array, i.e.,  $F = \{f_1, f_2, ..., f_k\}$ , where  $f_i$  is a complete vehicle route with opened depots inserted at its two ends. We also provide related information on the effects of vehicle and traveling on the objectives, such as starting/returning load, type of vehicle, traveling cost, time, and client waiting time in the attribute array, similarly to the solution representation proposed by Leng et al. [4].

Furthermore, a population-based search is used in our framework, so individuals in the population are constructed by randomly selecting clients to form a "super-client" (i.e., set of clients) assigned to each vehicle. Each vehicle is randomly assigned to the depot if only constraints (4)–(23) are met.

#### 4.1.2. Applied Operators

In this section, we provide domain-specific operators, as used by Leng et al. [2] for SOMs and Leng et al. [4] for biobjective models. Here, we first study the MORLCLRP by minimizing total cost, service duration, and client waiting time. Therefore, corresponding modifications are conducted in the applied operators by calculating the three objectives. However, we also follow the classification of Leng et al. [4], i.e., mutational operator (Mu), nondominated local search (NDLS), and dominated local search (DLS). Mu provides randomness to avoid local optima, NDLS produces many nondominated solutions, and DLS accelerates the process of searching approximate Pareto solutions by finding the solutions dominating parents. For iterations, DLS can provide high performance in short-term iterations; Mu plays a role in escaping local optima in medium-term iterations; and, NDLS performs best for nondominated solutions in long-term iterations. The corresponding design can be seen in Leng et al. [4].

## 4.1.3. General Structure of MOEAs for MORLCLRP

Algorithm 1 is the proposed general framework of MOEAs for the MORLCLRP. The proposed framework first creates an initial population (*Pop*) made of feasible random chromosomes. Afterwards, necessary parameters are set for the MOEAs, if needed. A main loop is then performed, stopping when the maximum number of generations ( $max_{gen}$ ) is reached.

In each generation, a Mu is randomly selected for the chosen individual (if random  $< p_m$ ) to obtain a child solution, then a local search operator is randomly chosen to optimize the obtained solution, which is merged into the parent population. Afterwards, if the size of the merged population is larger than N, the elitism selection strategy is applied to survive the best N individuals for the next generation. The algorithm ends when the main loop stops, returning the survived *Pop*.

Algorithm 1 General framework of MOEAs for LCLRP
<b>Input:</b> <i>Pop</i> , <i>max</i> <sub>gen</sub> , etc.
<b>Output:</b> child population ( <i>Pop</i> )
1: Generate corresponding parameters of meta-heuristics
//Main loop
2: Repeat
3: <i>i:=i</i> +1
4: <b>for</b> each solution in <i>Pop</i> <b>do</b>
// Mutation
5: <b>if</b> random< $p_m$ <b>then</b>
6: <b>Mu:</b> randomly choose a mutational operator
7: Obtain a child <i>C</i>
8: end if
// Local search
9: Local search: randomly select an operator from NDLS
10: and <b>DLS</b> to optimize Child <i>C</i> /Parent <i>P</i> (if random> <i>pm</i> )
11: Obtain a new solution <i>CC</i>
12: end for
13: Obtain child population <i>CP</i>
14. <b>Merge population:</b> <i>All=</i> [ <i>Pop,CP</i> ]
15: <b>Update of solutions:</b> apply the environmental selection of
meta-heuristic to generate the Pop into next generation
16: Stopping criteria: if stopping criteria is satisfied, then
stop and output <i>Pop</i> . Otherwise go to Step 3.
17: <b>until</b> $i = max_{gen}$

#### 4.2. Framework of MOHH

The proposed algorithm first creates an initial population (*Pop*) made of feasible random chromosomes, with certain acceptance criteria, selection strategies, and LLH parameters generated if needed. Afterwards, a main loop is implemented, stopping when the maximum number of iterations (i.e., *Max*<sub>iter</sub>) is reached.

For each main loop iteration, a promising MOEA is chosen by one selection strategy (i.e., SR, MAB, CF, and QS) to transform the solution space of the proposed problem, and *Cpop* is obtained. Afterwards, an acceptance criterion (i.e., AM, GDA, and LA) is performed to accept *Cpop* if the requirements of the selection strategy are met. The algorithm ends when the main loop stops,

returning the approximate Pareto solutions. We also provide an archiving strategy by storing  $5 \times N$  nondominated solutions where the nondominated sorting and crowded distance in NSGA-II [87] are used. An overview of the pseudocode of the proposed MOHH is given in Algorithm 2.

The main differences between the algorithm in this study and that of our previous work could be identified that the credit assignment (reward value) in this paper concerns D-mtrix (D), generational distance (GD), and population dominance (PD) instead of hypervolume (HV), PD, and Spacing. The main reason is that the difference between parent and child population can be identified by the first three. HV and Spacing only represent the characteristics of the child population.

<sup>//</sup> Initialization

- 1: *Citer*:=0 (Current iteration)
- 2: Parameter setting: parameters in Hypervolume, Maxiter, N, etc.
- 3: Generate population Pop
- 4: Calculate the multiobjective fitness

// Main loop

5: Cpop=Pop; Ppop=pop; ArPop=Ø;

## 6: Repeat

- 7: Citer:=Citer+1
- 8: // High-level selection strategy
- 9: Apply SR/MAB/CF/QS to select the promising LLH op
- 10: // Low-level heuristics
- 11: **Apply** the selected *opth* MOEA to generate *Cpop*
- 12: // High-level acceptance criterion
- 13: **Apply** the selected acceptance criterion (GDA, LA, and AM) to determinate the *Pop* of next iteration
- 14: // Archive population (if needed)
- 15: **Save** all nondominated individuals into *ArPop*;
- 16: **if** number of individuals in *ArPop* is larger than 5×*N* **then**
- 17: **Apply** environmental selection of NSGA-II to remove the individuals with much more crowded
- 18: end if
- 19: **until** Citer=Maxiter
- 20: if ArPop then
- 21: **Apply** environmental selection of NSGA-II to remove the 4×*N* individuals with much more crowded
- 22: Pop =ArPop
- 23: end if
- 24: return Pop

# 4.3. Heuristic Selection Strategies

In this study, well-known and modified heuristic selection approaches are used as selection MOHH components for choosing the appropriate MOEA at each iteration/stage.

## 4.3.1. Simple Random

For SR, a random LLH (MOEA) is simply chosen at each iteration. It is usually used as a baseline for comparison against the learning hyper-heuristic methods.

#### 4.3.2. Fitness Rate Rank Based Multi-Armed Bandit (FRR-MAB)

FRR-MAB is an upper confidence bound algorithm. In MAB, the key for a successful algorithm is to find a good trade-off between exploration and exploitation, i.e., the best LLH (exploitation) or other LLH (exploration). In this mechanism, there exists a credit assignment and operator selection.

In the process of FRR-MAB for MOM, the population dominance relationship is applied to calculate the reward of each LLH. Afterwards, the reward is stored in a sliding window, organized as a first-in-first-out mechanism. This is because, in dynamic environments, the performance of an operator in a very early stage may be irrelevant to its current performance [54]. Normalization is then applied to calculate the empirical quality estimate, and a confidence interval is obtained by the usage of operators. Finally, the credit value of each LLH is achieved. The LLH with a maximum credit value is selected for the next iteration.

#### 4.3.3. Choice Function

The CF heuristic selection method was first used by Maashi et al. [64] to tackle a series of continuous optimization benchmarks. In their CF, a two-stage ranking scheme calculates performance value, with CPU seconds elapsed since the last call to a low-level (meta) heuristic applied as the confidence internal. The final score (or value) for a given MOEA:

$$CF(llh) = \alpha f_1(llh) + f_2(llh), \ llh \in H$$
(29)

where,  $\alpha$  is a positive parameter; *H* is the pool of LLHs; *llh* is the index of an MOEA; *f*<sub>1</sub>/*f*<sub>2</sub> is the performance value(intensification)/elapsed CPU time (in seconds) since the time *llh* was last called (diversification). In this paper, we use a simple method to calculate the reward value of *llh* at *t* iteration using D-matrix, generational distance (GD), and population dominance (PD):

$$RV_{llh,t} = D \times GD \times PD \tag{30}$$

where, *D* is the difference in covered area size between parent front and child front; *GD* is the generational distance; and *PD* is the nondominated reward in FRR-MAB. The LLH with the highest *CF* value is preferentially selected.

#### 4.3.4. Quantum-Inspired Selection

The QS heuristic selection method proposed in our previous work [4] uses a qubit chromosome as a probabilistic representation instead of binary and numeric ones. Population dominance was used to judge the angle direction, and a comprehensive function that merges PD, HV, and spacing was used to reward the promising *llh*. A q-gate was introduced to update the state of each qubit in individual and heuristic space. The  $\beta$  value of each LLH reflected the exploitation of an algorithm, and probability matching (roulette selection) was used to explore the solution spaces to avoid local optima. In this paper, for fair comparison, we use the same reward function in CF to reward the promising *llh*. For the corresponding design of QS, please see Leng et al. [4].

## 4.4. Move Acceptance Methods

Move acceptance methods play a key role in MOHHs by deciding to accept or reject candidate solutions produced by the chosen LLH. There are several simple and elaborate acceptance strategies (AS) used as part of MOHHs. Maashi et al. [64] initially applied AM as an AS and after developed two much more efficient ASs: i.e., GDA and LA [61] which use D-matrix to provide the dominant relationship between parent and child populations. These ASs are usually used for MOHH-II modules, which utilize MOEAs as LLHs. Elitism selection strategy can also be ASs, known as best acceptance. Leng et al. [4] and Li et al. [60,83] used nondominated sorting and crowding distance

derived from NSGA-II [87] to select promising individuals for the next iteration. In this paper, we consider three well-known ASs: i.e., AM, GDA, and LA, as they performed well in Li et al. [60], Maashi et al. [61,64], and Leng et al. [4]. It is worth noting that the first two performs after normalizing the dimension of three objectives in this paper.

## 5. Computational Evaluation

#### 5.1. Implementation Aspects and Configuration of Parameters

The MOHH was coded in parallel in MATLAB 2018a (v9.4.0.813654) using a 4.0 GHz Intel Core i7-6700K with 12 GB of RAM and running Windows 10; it is embedded in the CLOR tool implemented by the MATLAB platform (available via email to the authors).

We followed the parameter configurations suggested by Leng et al. [4]. Here, the maximum iteration (*max*<sub>iter</sub>) was 100 for the outside loop. The maximum number of generations depended on the size of each instance:

$$Max_{gen} = \alpha \times (N_c + N_d + V_{max}) \tag{31}$$

where  $N_c(N_d)$  is the number of clients (depots); and  $V_{max}$  is the maximum number of vehicles with minimum capacity. Multiplier  $\alpha$  depended on the size of the instance and objectives, and was 45 for 20 clients, 57 for 30 clients, 68 for 40 clients, and 80 for 50 clients.

The size of the population and archiving population were set at 100 and 500, respectively. The mutation rate  $p_m$  was set to default (0.35), as per Leng et al. [4]. For parameters in the acceptance criteria, we applied the default values from Maashi et al. [61] and Leng et al. [4], i.e., rain speed ( $S = 3 \times 10^{-4}$ ) and C length (L = 5).

For the test suite, we applied the instances used by Leng et al. [4] for the biobjective RLCLRP. We generated other sets to analyze the effects of each speed zone area on key performance indicators, as described in Section 5.2.

#### 5.2. Performance Metrics

We utilized three well-known performance metrics to evaluate the performance of algorithms and problems by performing ten runs: inverted generated distance (IGD), HV, and the ratio of nondominated individuals (RNI).

The *IGD* describes the quality and uniformity of an approximate Pareto front (AF) by measuring the distance between AF and real Pareto front (PF), the smaller the *IGD* value, the better the distribution and convergence. The *HV* measures the covered space size between the *AF* and a reference point. The larger the *HV* value, the better the diversity and distribution. The last one is simply measured by putting together the nondominated solutions found by algorithms and the ratios between non-dominated solutions are reported. The larger the *RNI*, the better quality of *AF*.

However, as the MORLCLRP was first solved in this paper, if *PF* was unknown, we used all *AF* to form *PF*. The set *PF* was, in fact, an approximation of the real front. It is worth noting that the first two metrics were calculated after normalization.

The first hypothesis of this paper is that the use of hyperheuristics leads to better results than traditional MOEAs. The second is to analyze the effects of domain-specific parameters on the above three quality indicators. Considering these, our experiments were guided by the following research questions: (1) How do the results produced by the MOHH compare with traditional algorithms? and (2) what are the behaviors of the problem parameters affecting quality indicators?

To answer the aforementioned questions, we first applied eight MOEAs used in related work: nondominated sorting genetic Algorithm-II (NSGA-II) [87], strengthen Pareto evolutionary algorithm2 (SPEA2) [88], bi-goal evolution (BiGE) [89], nondominated sorting and local search (NSLS) [90], grid-based evolutionary algorithm (GrEA) [91], indicator-based evolutionary algorithm (IBEA) [92], NSGA-III [93], region-based selection in evolutionary-II (PESA-II) [94], then applied the three best-performing MOEAs as LLHs for MOHHs, followed by the next three well-performing MOEAs as LLHs. The results were evaluated using quality indicators. From the general questions, we will provide purposes in each experiment.

# 5.3. Efficiency of MOHH

Although Kumari and Srinivas [59] stated that the efficiency of general framework methods is inferior to problem-specific solvers incorporating domain-specific knowledge and fine-tuned, tailormade strategies, we compared our proposed MOHH and other MOEAs by analyzing the IGD and HV values by solving different instances. Tables 3 and 4 provide the mean IGD and HV values obtained by the eight MOEAs and two MOHHs with different LLHs, as mentioned above. Note that values in Tables 3 and 4 is multiplied 100 to save space.

As shown in Table 3, only SPEA2 and BiGE obtained the best mean IGD for L20-1 and L50-1, respectively. Similarly, although both obtained four third places, SPEA2 outperformed BiGE as the number of third places was larger than that for BiGE. As also seen in Table 3, QS1 using SPEA2, BiGE, and NSLS as LLHs achieved ten best IGD means compared with the first eight MOEAs and QS2 using NSGAII, NSGA-III, and PEAS-II as LLHs. Moreover, QS2 showed inferior performance to SPEA2 and BiGE, but was superior to others. Additionally, the MOHHs using archive methods outperformed the others as the mean IGD was the lowest among all algorithms, QS1a obtained nine best mean IGDs, whereas QS2a only achieved three.

Set	BiGE	GrEA	IBEA	NSGAII	NSGA-III	NSLS	PEAS-II	SPEA2	QS1	QS2	QS1a	QS2a
L201	2.68	3.41	4.66	2.65	3.05	2.44	2.65	2.12	2.24	2.43	1.72	1.64
L202	1.35	2.09	2.73	1.38	1.55	1.04	1.22	1.00	0.98	1.23	0.49	<u>0.47</u>
L203	1.67	2.64	4.08	1.46	2.14	1.36	1.43	1.33	1.28	1.39	0.82	0.75
L301	2.29	4.68	5.95	2.64	3.15	2.40	2.77	2.23	1.49	1.67	0.70	0.72
L302	3.381	5.01	6.50	3.51	4.34	3.39	3.46	3.380	3.17	3.90	2.20	2.30
L303	4.52	5.87	7.36	5.30	4.53	4.66	4.94	3.796	3.01	3.795	2.09	2.81
L401	2.37	3.73	4.36	3.54	3.23	2.90	2.67	2.61	2.36	2.87	<u>1.17</u>	1.29
L402	2.59	3.18	3.48	4.13	3.25	2.99	2.93	2.86	2.48	2.84	<u>1.17</u>	1.32
L403	2.57	4.10	4.93	3.77	3.48	3.01	2.84	2.83	2.57	2.98	1.336	1.338
L501	3.61	5.06	5.89	6.30	4.76	5.60	5.23	4.33	4.10	4.43	1.71	2.15
L502	3.79	5.74	6.76	5.31	5.19	4.17	5.67	3.50	3.18	3.95	1.94	2.10
L503	2.99	3.72	4.19	4.82	3.47	3.85	3.72	3.34	2.98	3.68	<u>1.67</u>	1.88
f/s/t	1/4/2	0/0/0	0/0/0	0/0/0	0/0/0	0/0/2	0/0/0	1/4/6	10/2/0	0/2/2	9	3

Table 3. Mean IGD values of 12 approaches for instances.

Note: bold numbers indicate the minimum values; italic are the second minimum values; and numbers with underline are the third minimum values; the last two columns are MOHH with archiving methods.

Table 4. Mean HV values of 12 approaches for instances.

Set	BiGE	GrEA	IBEA	NSGAII	NSGA-III	NSLS	PEAS-II	SPEA2	QS1	QS2	QS1a	QS2a
L201	2.1467	2.1389	2.1199	2.1422	2.1520	2.1555	2.1401	2.1680	2.1647	2.1594	2.1699	2.1769
L202	0.8204	0.8163	0.8161	0.8184	0.8105	0.8207	0.8180	0.8221	0.8218	0.8197	0.8251	0.8252
L203	1.7113	1.7094	1.7007	1.7059	1.7016	1.7091	1.7021	1.7097	1.7105	1.7083	1.7151	1.7154
L301	1.0995	1.0695	1.0670	1.0732	1.0519	1.0812	1.0555	1.0988	1.1603	1.1511	<u>1.1678</u>	1.1657
L302	2.8058	2.7526	2.7432	2.8109	2.7704	2.7786	2.7822	2.7596	2.8053	2.7684	2.8287	2.8139
L303	1.7902	1.7889	1.7584	1.6971	1.7939	1.7467	1.7393	1.8375	1.9062	1.8478	1.9222	1.8688
L401	2.0993	2.0970	2.0870	2.0676	2.0394	2.0973	2.0661	2.0970	2.1067	2.0710	2.1468	2.1396
L402	2.3967	2.4103	2.4153	2.3796	2.3799	2.4285	2.3836	2.4312	2.4457	2.3981	2.4919	2.4845
L403	2.9053	2.8752	2.8740	2.8989	2.8676	2.9062	2.8898	2.9091	2.9145	2.8898	2.9560	2.9553
L501	2.2779	2.2738	2.1761	2.2312	2.2345	2.2130	2.1681	2.3257	2.3613	2.3190	2.4770	2.4676
L502	2.2264	2.1440	2.1580	2.1760	2.1605	2.2007	2.0440	2.2392	2.3101	2.2517	2.3589	2.3444
L503	2.3465	2.3506	2.3081	2.3510	2.3456	2.3945	2.3042	2.3914	2.4076	2.3388	2.4609	2.4466
f/s/t	1/2/0	0/0/0	0/0/0	1/0/0	0/0/0	0/1/3	0/0/0	2/3/6	8/3/1	0/3/2	9	3

Note: bold numbers indicate the minimum values; italic are the second minimum values; and numbers with underline are the third minimum values; the last two columns are MOHH with archiving methods.

Table 4 displays the mean HV for all instances. Similarly, the performances of BiGE and SPEA2 were the best among the first eight MOEAs, especially SPEA2, which achieved two first places, three second places, and six third places. NSGAII also obtained one first place. Looking at the performance

using the hyper-heuristic framework, QS1 claimed first place by achieving eight first places, three second places, and one third place in 12 instances; QS2 showed lower performance achieving only three second places and two third places. The archiving method thus improved the performance of MOEAs as stated in Zhang et al. [95].

Figure 1 is a boxplot of hypervolumes obtained by performing 12 MOEAs. The following conclusions were drawn: (1) A single MOEA cannot outperform all other MOEAs in all instances, e.g., BiGE for L50-1 and L50-3 and SPEA2 for L50-3; (2) BiGE, SPEA2, and NSLS were superior to others, and NSGA-II, NSGA-III, and PEAS-II outperformed GrEA and IBEA in most instances. We used the scoring system [96] to classify the first eight MOEAs, and found SPEA2(213) < BiGE(180) < NSLS(146) < PEAS(93) < NSGAII(92) < NSGAIII(87) < GrEA(79) < IBEA(46), where A(*a*) < B(*b*) indicates that the score *a* of A is larger than the score *b* of B and thus A dominates B. Therefore, QS1 used SPEA2, BiGE, and NSLS as LLHs and QS2 used PEAS, NSGAII, and NSGAIII as LLHs. (3) The use of hyperheuristics (i.e., QS1) outperformed others without the hyper-heuristic framework, however, QS2 was inferior to the top three MOEAs, but outperformed the other five MOEAs, suggesting that the configuration of the LLH pool is extremely significant in designing MOHHs. (4) The archiving method can significantly improve the performance of MOEAs, as shown in Tables 3 and 4, and Figure 1; therefore, the following experiments (except Section 5.4) used the archiving method to obtain better Pareto solutions. In brief, the hyperheuristic in this paper effectively improved performance of MOEAs by assimilating the essence and rejecting the dross of MOEAs.



Figure 1. Boxplot of hypervolume of 12 MOEAs.

### 5.4. Efficiency of Pairs in MOHH

This section compares the performance of each pair described in Section 4.2 and 4.3. Four selection strategies and three acceptance criteria formed 12 pairs, which were used to optimize the randomly generated instances. Figures 2 and 3 are boxplots of the IGD and HV values, respectively. For an easy plot, we used the logogram of GDA (GD) and FRR-MAB (FM).



Figure 2. Boxplot of IGD values of 12 pairs.



Figure 3. Boxplot of hypervolume values of 12 pairs.

As shown in Figure 2, the performance of GDA was inferior to the others, even though they shared the same SSs. The performance of LA was superior to that of AM in most instances when sharing the same SS. However, different SSs sharing the same AC demonstrated different performances. For example, QSLA outperformed the others in L202, L401. A similar pattern for all pairs can be seen in Figure 3. As comparing the performance of all pairs can be difficult, we used a scoring system (see Section 5.3), with results shown in Table 5. QSLA, SRLA, QSAM, FMLA, and CFAM, were the top five pairs. The score of each SS with GDA was equal to zero, indicating that GDA was the worst AC. Conversely, LA was the best AC, and was 20 scores ahead compared with AM. The order of performance was QSLA > SRLA > QSAM > FMLA > CFAM > SRAM > FMAM > CFLA. The order of performance of AC when neglecting SS was LA > AM > GDA. The order of performance of the four SSs when neglecting AC was QS > CF > SR > FM. In the following experiments,

we randomly selected one of the top three pairs in each run, i.e., QSLA, SRLA, and QSAM, to analyze the effects of domain-specific parameters on IGD and HV values.

	SR	FRR-MAB	QS	CF	Total
AM	110	107	126	115	458
LA	130	116	133	99	478
GDA	0	0	0	0	0
Total	240	223	259	241	NA

Table 5. Scores of each HLH strategy.

# 5.5. Effect of Clients and Depots Locations

As each speed zone has a fixed speed, which is a significant parameter in FCCE, the locations of clients and depots are extremely important for determining which to serve. In this test suite, we randomly generated 16 instances with different depot and client locations, i.e., CnDm,  $n \in \{1,2,3,R\}$  and  $m \in \{1,2,3,R\}$ , where, *R* indicates the location of depots/clients is randomly located within a speed zone. Tables 6 and 7 are the mean IGD and HV values of 16 pairs for each instance. Note that values in Tables 6 and 7 is multiplied 100 to save space.

Set	CRDR	CRD1	CRD2	CRD3	C1DR	C1D1	C1D2	C1D3	C2DR	C2D1	C2D2	C2D3	C3DR	C3D1	C3D2	C3D3
L201	17.6	23.6	16.3	12.5	4.8	9.2	6.5	8.3	3.6	10.6	5.0	3.7	17.0	36.3	24.4	16.2
L202	36.0	41.0	36.1	34.8	4.4	8.5	5.8	9.2	13.3	19.4	16.5	15.4	25.2	29.7	25.5	22.7
L203	22.1	23.1	27.4	21.4	4.5	7.2	6.8	12.6	13.9	13.3	11.2	12.2	19.1	21.5	17.3	16.0
L204	14.4	22.8	17.0	16.6	2.7	6.9	5.9	7.4	6.6	12.4	7.5	10.0	12.6	21.7	15.5	14.0
L301	13.8	19.6	20.6	14.6	4.5	4.2	4.3	8.7	4.8	7.8	5.9	10.0	11.4	32.1	20.5	11.2
L302	17.2	40.5	17.6	17.9	3.2	7.3	4.4	7.2	11.7	14.8	10.1	13.1	27.8	26.9	21.1	16.7
L303	16.7	19.1	16.1	16.3	4.9	6.4	4.8	7.2	7.9	10.9	8.5	9.5	11.0	14.8	10.7	11.0
L304	31.4	36.2	29.4	26.7	4.0	6.6	4.3	11.1	13.1	16.0	11.5	13.0	32.2	36.0	31.4	27.7
L401	9.4	24.3	13.4	10.9	4.5	10.2	4.7	9.2	5.2	14.3	8.8	9.8	7.2	21.5	12.9	6.2
L402	11.6	22.6	13.6	8.8	3.3	6.3	7.1	6.2	6.2	14.1	7.1	7.2	10.3	20.2	9.9	7.1
L403	11.1	23.8	14.0	11.6	3.0	4.3	4.7	10.1	4.1	16.1	8.7	7.6	12.6	25.7	13.4	7.8
L404	9.8	21.2	9.4	12.1	5.1	2.5	7.3	5.5	10.6	10.3	12.1	9.8	11.4	15.0	15.3	10.4
L501	19.7	26.0	19.9	20.0	3.2	1.9	7.3	11.3	10.5	14.4	9.7	10.9	17.4	31.2	19.3	11.8
L502	10.6	29.4	17.4	10.8	2.2	12.8	7.1	7.9	3.1	7.0	7.6	5.8	10.8	28.6	17.8	14.0
L503	14.2	25.8	16.8	13.1	4.5	1.6	3.3	8.6	5.7	17.8	9.4	8.7	13.5	29.3	17.1	13.8
L504	19.2	34.5	22.8	16.9	2.2	2.8	8.0	6.4	11.0	22.8	14.4	11.2	15.2	31.4	22.2	15.0

Table 6. Mean IGD values of 16 pairs.

Set	CRDR	CRD1	CRD2	CRD3	C1DR	C1D1	C1D2	C1D3	C2DR	C2D1	C2D2	C2D3	C3DR	C3D1	C3D2	C3D3
L201	15.0	12.7	17.6	19.9	25.2	23.0	25.1	23.5	27.4	22.3	25.9	27.4	16.3	6.8	12.4	16.9
L202	9.6	7.0	9.4	10.8	32.8	27.6	30.7	31.9	24.2	19.5	22.6	24.5	15.4	12.1	14.5	17.2
L203	10.9	9.8	7.0	11.4	25.7	24.0	25.7	21.2	14.3	14.6	17.1	16.4	12.8	11.0	13.2	14.6
L204	18.8	12.9	17.2	18.6	32.4	28.5	29.4	29.9	28.4	23.2	27.2	28.3	22.4	16.2	20.5	23.8
L301	19.6	14.9	15.5	19.4	28.3	29.5	28.2	25.7	31.1	27.9	30.5	26.6	23.0	12.1	17.4	23.8
L302	19.1	8.2	17.7	17.6	32.5	29.1	32.2	30.1	22.2	20.9	23.7	21.4	12.6	12.7	16.1	19.7
L303	18.2	16.2	18.1	20.3	30.6	29.6	31.1	32.4	25.2	22.7	25.2	27.2	23.6	20.6	22.9	25.2
L304	0.8	0.5	1.6	2.1	17.1	15.0	16.6	16.5	7.1	5.7	7.0	7.2	0.2	0.0	0.1	0.9
L401	24.8	14.1	21.2	23.7	34.0	25.6	31.4	33.4	29.1	21.5	25.0	30.4	27.9	16.7	22.3	30.1
L402	24.5	17.1	23.0	28.0	35.4	30.3	29.1	32.4	28.4	21.9	27.0	29.1	24.9	17.6	25.7	28.6
L403	22.2	15.0	20.3	22.5	31.9	31.1	32.3	28.4	29.4	19.8	24.3	28.9	22.0	13.6	21.6	26.2
L404	33.0	22.9	32.1	33.9	40.0	40.1	35.6	42.3	31.8	30.7	29.0	34.6	32.3	27.8	27.0	33.1
L501	16.6	12.4	15.7	16.5	29.5	29.0	23.1	22.9	20.8	17.7	20.9	22.8	16.1	9.2	15.7	20.9
L502	18.9	9.6	14.4	18.3	27.9	19.8	22.6	23.7	25.3	22.4	22.1	23.4	18.4	10.3	14.9	16.5
L503	20.8	14.2	19.2	22.1	30.0	31.8	32.0	30.7	28.6	19.4	25.2	27.1	21.6	12.6	19.4	22.0
L504	22.2	12.9	19.1	24.1	37.8	34.5	30.8	32.7	27.7	19.2	24.1	28.1	25.1	14.4	19.8	25.4

Table 7. Mean HV values of 16 pairs.

Looking at Table 6, in most instances, the mean IGD values of C1DR, C1D1, C1D2, and C1D3 were the top four allocation strategies. The worst allocation strategies were C3D1, CRD1, C3D2, and CRD2. Similar characteristics can also be derived from Table 7. We analyzed the RNI indictors in a "real" Pareto set, and found the sequence: C1D1(29.6%) > C1DR(21.6%) > C1D3(19.5%) > C1D2(13.3%) > C2D3(6.0%) > C2D2(4.0%) > C2DR(3.1%) > C3D3(1.6%) > CRD3(5.2‰) > C2D1 (2.9‰) > C3DR(2.1‰) > CRD1(0.733‰) > CRD2(0.729‰) > C3D1(0.22‰) > CRDR(0.13‰) > C3D2 (0.056‰), where the numbers in parentheses are RNI values. Although the costs of depots in zone 1 were highest, the RNI was the largest among the 16 pairs. The RNI values of C1DR, C1D3, and C1D2 were also higher than other pair. In fact, the RNI values did not represent the priority order among the 16 pairs.

## 5.6. Effect of Fleet Composition

In this section, we analyzed the effects of fleet composition on Pareto front using IGD, HV, and RNI indicators. The instance sets were the same as in Section 5.3. Table 8 shows the mean IGD, HV, and RNI values for each pair. The "HF" represents heterogonous fleet. Note that values in Table 8 is multiplied 100 to save space.

Instance		IG	D			Н	V			RNI				
Instance	L1	L2	М	HF	L1	L2	М	HF	L1	L2	Μ	HF		
L201	36.70	4.27	1.72	0	0.04	18.46	21.94	22.62	0	13.45	8.07	100		
L202	15.06	0.94	3.83	0	2.43	8.10	7.57	8.26	0	12.40	1.24	100		
L203	12.01	0.83	2.99	0	10.18	17.78	16.88	17.94	0	36.01	5.68	100		
L301	12.64	0.81	3.37	0	6.15	12.85	11.87	12.96	0	33.06	5.78	100		
L302	28.18	2.37	1.59	0	10.45	28.80	29.64	30.80	0	26.80	21.97	100		
L303	26.37	1.76	2.67	0	3.20	19.50	18.91	20.20	0	2.62	0.52	100		
L401	24.90	3.17	1.51	0	16.68	31.34	31.90	35.03	0.51	4.33	10.90	100		
L402	33.68	4.59	1.13	0	4.85	20.63	24.70	25.70	0	8.11	4.28	100		
L403	32.74	3.34	1.26	0	9.33	27.24	29.20	29.96	0	18.18	31.37	100		
L501	27.56	3.44	1.25	0	16.03	31.90	36.56	37.74	0.02	3.06	3.78	100		
L502	47.84	9.33	1.08	0	0	18.74	26.09	26.60	0	1.41	15.71	100		
L503	26.50	4.47	2.77	0	14.21	28.66	30.08	32.06	4.96	4.13	1.45	100		

Table 8. Mean IGD, HV, and RNI values of each fleet composition.

As shown in Table 8, the mean IGD, HV, and RNI values of HF for each instance were better than those of L1, L2, and M. The "real" Pareto front of each instance was the same as fronts of HF. As vehicle parameters were different, the preference of distribution in each instance was also different. For example, type L2 vehicles were preferred in L20, L30, L402, and L503, whereas type M vehicles were preferred in L401, L403, L501, and L502. Therefore, as the size of instance increases, a type M vehicle will be preferred. However, the total RNI indicator of L1, L2, and M was 48.8%, lower than 50%, which demonstrates that a composited HF can reduce logistics costs, service duration, fuel consumption, carbon emissions, and client satisfaction. From the perspectives of IGD and HV values, the Pareto front of HF dominated the fronts of L1, L2, and M after removing the duplicating individuals. Figure 4 is a boxplot of IGD, HV, and RNI for each fleet composition.





Figure 4. Boxplot of IGD, HV, and RNI values of each fleet composition.

## 5.7. Effect of Zones Area

Here, we estimated the effects of speed zone area on IGD, HV, and RNI values. The instances were randomly generated. Table 9 lists the IGD, HV, and RNI of each instance, where EQ/Do/TF/FF/ORI represent equal/two-fold/three-fold/four-fold/original from Koc et al. [1]. Note that values in Table 9 is multiplied 100 to save space.

Figure 5 is a boxplot of IGD, HV, and RNI values for each area ratio. As seen in Table 9, as the ratio of the zone area increased, performance also increased, especially FF. The ratio of FF obtained the best performance, except for IGD/RNI values of L501 and HV values of L401, L501, and L502. The ratio of the ORI area zone was zone1 (9%): zone 2 (25%): zone 3 (64%). An ideal ratio for speed zone area for specific instances was shown to exist. As shown in Table 9, ORI outperformed FF in L501, which indicates that the best ratio may exist between ORI and FF; however, the best ratio is extremely hard to find as infinite pairs exist and it may change with the nature of instances.

Table 9. Mean IGD, HV, and RNI values of each area ratio.

Set IGD								HV			RNI					
Set	EQ	Do	TF	FF	ORI	EQ	Do	TF	FF	ORI	EQ	Do	TF	FF	ORI	
L201	13.53	6.23	3.14	0.36	3.23	6.64	11.54	13.51	15.24	13.44	0	0	4.17	86.67	9.17	
L202	8.77	8.49	1.12	0.52	1.83	8.42	8.62	11.32	11.74	11.07	0.52	2.61	15.65	61.91	19.30	
L203	6.44	4.59	3.58	0.08	4.25	20.36	22.61	23.84	26.38	22.94	1.18	2.19	0	94.95	1.68	
L301	11.52	3.35	1.96	0.32	1.64	9.69	12.72	13.25	15.09	13.29	0	0	12.80	61.20	26.01	
L302	7.29	1.84	1.07	0.66	0.98	22.05	26.92	27.65	28.02	27.45	0.59	10.42	11.68	48.85	28.46	
L303	18.99	4.51	3.65	0.14	3.57	10.24	19.43	20.11	23.43	20.21	0	0	3.16	92.21	4.62	
L401	7.70	2.63	1.11	0.59	1.33	16.11	20.24	20.77	20.85	22.04	0	0.66	27.21	40.48	31.65	
L402	17.83	7.00	2.35	0.03	5.82	12.68	18.15	21.75	23.75	19.50	0	0	2.12	97.88	0	
L403	10.29	5.16	3.66	0.11	3.46	19.90	22.92	24.41	28.88	24.58	0	0	0.20	97.49	2.31	
L501	10.54	7.25	1.41	1.29	0.31	15.72	19.55	24.24	24.42	26.73	0	0	4.90	22.67	72.43	
L502	6.79	4.00	0.95	0.83	0.86	19.61	21.33	23.87	23.84	24.21	0.06	0.63	19.19	48.45	31.67	
L503	13.37	6.41	1.34	0.55	1.05	15.60	21.33	25.31	25.96	25.37	0	0	12.28	58.93	28.79	



Figure 5. Boxplot of IGD, HV, and RNI values of each area size.

#### 5.8. Management Implications

From a management point of view, several management implications can be obtained from the results. Section 5.5 analyzed the joint effects of depot and client location. The depot and client preference in each zone are shown in Figure 6. From the perspective of depots in each zone, the clients in zone 1 were preferred for the DR, indicating that the randomly located depots (zones 1, 2, and 3) should be given preference to serve the clients in zone 1 instead of other zones. All depots located in each zone preferred to serve clients in zone 1, indicating that the depots in D2 and D3 preferred to serve clients in C1 instead of C2 and C3, respectively. From the perspectives of clients, the first three groups (i.e., CR, C1, and C2) preferred DR for service rather than the depots in D1 and D2. However, the clients in zone 3 preferred the depots in zone 3 instead of DR. Moreover, the clients in CR, C1, C2, and C3 preferred depots in D3, D2, D3, and DR, respectively. The worst allocation strategy was clients in zone 3 served by depots in zone 1. The priority was: C1DR < C1D2 < C1D1 < C1D3 < C2DR < C2D3 < C2D2 < C3D3 < C2D1 < C3DR < CRDR < CRD3 < CRD2 < C3D2 < CRD1 < C3D1. The main reasons for these results are: (1) Zone-specific parameters: i.e., speed and geographical factors. Each zone had a different vehicle speed affecting FCCE amount, and the size of each zone was also another affecting parameter; (2) Domain-specific parameters. The cost of depots in each zone was the most significant factor determining the selection of depots to serve clients. Therefore, logistics companies should analyze the effect of client and depot distribution on Pareto solutions, as the calculation of costs, service duration, fuel consumption, carbon emission, and client satisfaction are based on client and depot locations.



Figure 6. Preference of clients and depots locations.

Section 5.6 analyzed the effects of fleet composition. From the results, we strongly conclude that the fleet composition consisted of HF can provided reduction in logistics costs, service duration, fuel consumption, carbon emission, and client satisfaction. Section 5.7 analyzed the effects of speed zone area, and we found that a best ratio range may exists for the speed zones and should be analyzed and determined, which is important for city planning decided by the economy and population development. Moreover, it is important for logistics enterprises to decide whether or not to authorize other logistics (such as local logistics companies) with a low entrust cost instead of building or renting the facilities (depots and vehicles) with a higher fixed cost.

In the experiments, we analyzed the efficiency of the proposed general framework, i.e., MOHH. Results showed that the proposed algorithms effectively tackled the proposed problem and outperformed several well-known MOEAs. Moreover, we analyzed the effects of domain-specific parameters, such as depot and client locations, fleet composition, and speed zone area on the Pareto solution indicators, i.e., IGD, HV, and RNI. We also provided several management and service suggestions to help reduce total costs (including FCCE cost) and service duration, as well as increase client satisfaction.

## 6. Conclusions

In this paper, we presented a MOHH algorithm to solve a MORLCLRP considering simultaneous pickup and delivery, hard time windows, and heterogeneous fleets. In the problem domain, we modeled a multiobjective mathematical formula, which simultaneously minimized service duration, client waiting time, and total logistics costs, where the latter was defined with respect to total costs of renting depots and vehicles as well as FCCE. In the algorithms, we proposed a MOHH, with simple random, fitness rate rank-based multi-armed bandit, choice function, and quantum-inspired selection as the high-level strategy, and all moves, great deluge, and late acceptance as acceptance criteria. Moreover, we provided a general framework of MOEAs used to perform comparative analysis.

In regard to the study aims, we: (1) verified the efficiency of the proposed algorithms; (2) comparatively analyzed the performance of 16 pairs; and (3) analyzed the effects of domain-specific parameters on performance indicators. The first experiment verified our proposed algorithms compared with eight well-known MOEAs. The second experiment showed that the performances of QS-LA, SR-LA, and QS-AM were the best among the 16 pairs. The third experiment evaluated the impact of problem parameters on Pareto solutions, and several conclusions could be obtained:

- Although method synthesis might promote the algorithm's performance, the strategy is significant to choose and monitor the performance of each method. Moreover, the LLHs have strong ability in effecting the whole performance, therefore, the analysis should be conducted before constructing the pool of low-level heuristics;
- The HLHs are important for the algorithm's performance, and the unmerited design may
  produce the poor performance than the simple random, such as FRR-MAB and GDA;
- The depot and client location has significant impacts on the logistics cost, client satisfaction, and service duration. Before determining the set of depots to open and the tracing of the routes, the joint effects of the depot and client location should be analyzed. In the context of this paper, the joint effect can be obtained: C1DR > C1D2 > C1D1 > C1D3 > C2DR > C2D3 > C2D2 > C3D3 > C2D1 > C3DR > CRDR > CRD3 > CRD2 > C3D2 > C3D1 > C3D1 (priority sequence). Moreover, we also analyzed the preference of client and depot in each speed zone, and we found that DR/D3 (C1→C2→C3→CR), D1/D2(C1→C2→CR→C3), CR/C2(DR→D3→D2→D1), C1(DR→D2→D1→D3), and C3(D3→DR→D2→D1), where A(B→C) indicates that the preference of A is B and C, and B is better than C;
- The fleet composition is another factor effecting the logistics network. From the perspectives of the results, the heterogonous fleet could always obtain better Pareto solutions;
- The zone area effects the depot and client location, and how to partition the speed zones (i.e., the best ratio of speed zones) and determine the depot and client location, to some extents, determine the logistics network. Therefore, the government and logistics companies should optimize the speed zone area for economic, environmental, and social effects.

However, as ours is a multiobjective model, the ratio of fuel consumption and CE cost is difficult to analyze. Moreover, although hyper heuristics are oriented from the generality level, the performance of LLHs significantly effects the whole performance, therefore the future work will focus on the development of MOHH-II, and may try to improve the performance of multiobjective hyperheuristics by designing more efficient high-level strategies. we may also consider the uncertainty related to the input parameters with fuzzy method [97], stochastic program [98,99], and practical application [6,100] to bring the problem closer to the reality.

**Author Contributions:** L.L. came up with the main idea of model and algorithm and carried out the simulations. Y.Z. helped with the structure and revisions of the paper. J.Z. and C.Z. investigated the state-of-the-art literature and pointed out some suggestions about literature review.

Funding: This research was funded by the National Natural Science Foundation of China grant numbers 61572438, 61873240, 61402409, National Natural Science Foundation of Zhejiang grant number Y19F030052 and

Science Technology plan project of Zhejiang grant number 2017C33224. This support is gratefully acknowledged. Thanks for the reviewers and editors who provided valuable comments and suggestions for revising this paper.

Conflicts of Interest: The authors declare no conflict of interest.

## References

- 1. Koc, C.; Bektas, T.; Jabali, O.; Laporte, G. The impact of depot location, fleet composition and routing on emission in city logistics. *Transp. Res. Part B Methodol.* **2016**, *84*, 81–102, doi:10.1016/j.trb.2015.12.010.
- Leng, L.; Zhao, Y.; Wang, Z.; Wang, H.; Wang, W. Shared mechanism based self-adaptive hyperheuristic for regional low-carbon location-routing problem with time windows. *Math. Probl. Eng.* 2018, 2018, 8987402, doi:10.1155/2018/8987402.
- 3. Zhang, C.M.; Zhao, Y.W.; Zhang, J.L.; Leng, L.L. Location and routing problem with minimizing carbon. Comp. *Integr. Manuf. Syst.* **2017**, *23*, 2768–2777, doi:10.13196/j.cims.2017.12.023.
- Leng, L.; Zhao, Y.; Wang, Z.; Zhang, J.; Zhang, C. A novel hyper-heuristic for bi-objective regional lowcarbon location-routing problem with multiple constraints. *Sustainability* 2019, *11*, 1596, doi:10.3390/su11061596.
- 5. Pourhejazy, P.; Kwon, O.K.; Lim, H. Integrating sustainability into the optimization of fuel logistics networks. *KSCE J. Civ. Eng.* **2019**, *23*, 1369–1383, doi:10.1007/s12205-019-1373-7.
- 6. Qin, G.; Tao, F.; Li, L. A vehicle routing optimization problem for cold chain logistics considering customer satisfaction and carbon emissions. *Int. J. Environ. Res. Public Health* **2019**, *16*, 576, doi:10.3390/ijerph16040576.
- 7. Pourhejazy, P.; Kwon, O. The new generation of operations research methods in supply chain optimization: A review. *Sustainability* **2016**, *8*, 1033, doi:10.3390/su8101033.
- 8. Drexl, M.; Schneider, M. A survey of variants and extensions of the location-routing problem. *Eur. J. Oper. Res.* **2015**, *241*, 283–308, doi:10.1016/j.ejor.2014.08.030.
- Ardekani, S.; Hauer, E.; Jamei, B. Traffic impact models. In the Traffic Flow Theory: A State-of-the Art Report; Federal Highway Administration Research and Technology, Washington DC, USA, 1996; Chapter 7; pp. 7– 24.
- 10. Bigazzi, A.; Bertini, R.L. Adding green performance metrics to a transportation data archive. *Transp. Res. Rec.* **2009**, *2121*, 30–40, doi:10.3141/2121-04.
- 11. Alwakiel, H.N. Leveraging Weigh-in-Motion (WIM) Data to Estimate Link-Based Heavy-Vehicle Emissions. Ph.D. Thesis, Portland State University, Portland, OR, USA, 2011. doi:10.15760.edu.247.
- 12. Lin, C.; Choy, K.L.; Chung, S.H.; Lam, H.Y. Survey of green vehicle routing problem: Past and future trends. *Expert Syst. Appl.* **2014**, *41*, 1118–1138, doi:10.1016/j.eswa.2013.07.107.
- 13. Demir, E.; Bektas, T.; Laporte, G. A comparative analysis of several vehicle emission models for road freight transportation. *Transp. Res. Part D Transp. Environ.* **2011**, *16*, 347–357, doi:10.1016/j.trd.2011.01.011.
- 14. Demir, E.; Bektas, T.; Laporte, G. A review of recent research on green road freight transportation. *Eur. J. Oper. Res.* **2014**, *237*, 775–793, doi:10.1016/j.ejor.2013.12.033.
- 15. Demir, E.; Bektas, T.; Laporte, G. The bi-objective pollution-routing problem. *Eur. J. Oper. Res.* **2014**, 232, 464–478, doi:10.1016/j.ejor.2013.08.002.
- 16. Poonthalir, G.; Nadarajan, R. A fuel efficient green vehicle routing problem with varying speed constraint (F-GVRP). *Expert Syst. Appl.* **2018**, *100*, 131–144, doi:10.1016/j.eswa.2018.01.052.
- 17. Kuo, Y. Using simulated annealing to minimize fuel consumption for the time-dependent vehicle routing problem. *Comput. Ind. Eng.* **2010**, *59*, 157–165, doi:10.1016/j.cie.2010.03.012.
- 18. Kazemain, I.; Aref. S. A green perspective on capacitated time-dependent vehicle routing problem with time window. *Int. J. Supply Chain Manag.* 2017, *2*, 20–38, doi:10.1504/IJSCIM.2017.10007322.
- 19. Mirmohammadi, S.H.; Tirkolaee, E.B.; Goli, A.; Dehnavi-Arani, S. The periodic green vehicle routing problem with considering of time-dependent urban traffic and time windows. *Int. J. Opt. Civ. Eng.* **2017**, *7*, 143–156.
- Andersson, H.; Hoff, A.; Christiansen, M.; Hasle, G.; Lokketanen, A. Industrial aspects and literature survey: Combined inventory management and routing. *Comput. Oper. Res.* 2010, 37, 1515–1536, doi:10.1016/j. cor.2009.11.009.
- 21. Koc, C.; Bektas, T.; Jabali, O. The fleet size and mix pollution-routing problem. *Transp. Res. Part B Methodol.* **2014**, *70*, 239–254, doi:10.1016/j.trb.2014.09.008.
- 22. Koc, C.; Bektas, T.; Jabali, O. The fleet size and mix location-routing time windows: Formulations and a heuristics algorithm. *Eur. J. Oper. Res.* **2016**, *248*, 33–51, doi:10.1016/j.ejor.2015.06.082.
- 23. Pitera, K.; Sandoval, F.; Goodchild, A. Evaluation of emissions reduction in urban pickup systems heterogeneous fleet case study. *Transp. Res. Rec.* **2011**, 2224, 8–16, doi:10.3141/2224-02.

- 24. Xiao, Y.Y.; Konak, A. The heterogeneous green vehicle routing and scheduling problem with time-varying traffic congestion. *Transp. Res. Part B Methodol.* **2016**, *88*, 146–166, doi:10.1016/j.tre.2016.01.011.
- Xiao, Y.Y.; Zhao, Q.H.; Falu, I. Development of a fuel consumption optimization model for the capacitated vehicle routing problem. *Comput. Oper. Res.* 2012, *39*, 1419–1431, doi:10.1016/j.cor.2011.08.013.
- 26. Shen, L.; Tao, F.; Wang, S. Multi-depot open vehicle routing problem with time windows based on carbon trading. *Int. J. Environ. Res. Public Health* **2018**, *15*, 2025, doi:10.3390/ijerph15092025.
- 27. Wang, S.; Tao, F.; Shi, Y. Optimization of location-routing problem for cold chain logistics considering carbon footprint. *Int. J. Environ. Res. Public Health* **2018**, *15*, 86, doi:10.3390/ijerph15010086.
- 28. Leng, L.; Zhao, Y.; Zhang, C.; Wang, S. Quantum-inspired hyper-heuristics for low-carbon location-routing problem with simultaneous pickup and delivery. *Comp. Integr. Manuf. Syst.* **2018**, in press.
- Zhao, Y.; Leng, L.; Wang, S.; Zhang, C. Evolutionary hyper-heuristics for low-carbon location-routing problem with heterogeneous fleet. *J. Control. Dec.* 2018, doi:10.13195/j.kzyjc.2018.0756.
- Wang, S.; Zhao, Y.; Leng, L.; Zhang, C. Research on low carbon location routing problem based on evolutionary hyper-heuristic algorithm of ant colony selection mechanism. *Comp. Integr. Manuf. Syst.* 2018, in press.
- Kan, Z.; Tang, L.; Kwan, M.P.; Zhang, X. Estimating vehicle fuel consumption and emissions using GPS big data. *Int. J. Environ. Res. Public Health* 2018, 15, 566, doi:10.3390/ijerph15040566.
- Xiao, L.; Dridi, M.; Hassani, A.H.E.; Fei, H.; Lin, W. An improved cuckoo search for a patient transportation problem with consideration of reducing transport emissions. *Sustainability* 2018, 10, 793, doi:10.3390/su10030793.
- Lee, s.; Hwang, T. Estimating emissions from regional freight delivery under different urban development scenarios. *Sustainability* 2018, 10, 1188, doi:10.3390/su10041188.
- Hwang, T.; Ouyang, Y. Urban freight truck routing under stochastic congestion and emission considerations. *Sustainability* 2015, 7, 6610–6625, doi:10.3390/su7066610.
- Rakha, H.; Ahn, K.; Moran, K.; Saerens, B.; Van de Bulck, E. Simple comprehensive fuel consumption and CO<sub>2</sub> emissions model based on instantaneous vehicle power. In Proceedings of the 90th Transportation Research Board Annual Meeting, Washington, DC, USA, 23–27 January 2011.
- Bandeira, J.; Carvalho, D.O.; Khattak, A.J.; Rouphail, N.M.; Coe; Coelho, M.C. A comparative empirical analysis of eco-friendly routes during peak and off-peak hours. In Proceedings of the Transportation Research Board 91st Annual Meeting, Washington, DC, USA, 22–26 January 2012.
- Bandeira, J.; Carvalho, D.O.; Khattak, A.J.; Rouphail, N.M.; Coelho, M.C. Generating emissions information for route selection: Experimental monitoring and routes characterization. *J. Intell. Transport. Syst.* 2013, 17, 3–17, doi:10.1080/15472450.2012.706197.
- Japanese Government Website. Available online: http://www.mlit.go.jp/common/000037099.pdf (accessed on 18 March 2019).
- Chen, C.; Qiu, R.; Hu, X. The location-routing problem with full truckloads in low-carbon supply chain network designing. *Math. Probl. Eng.* 2018, 2018, 6315631, doi:10.1155/2018/6315631.
- Mohammadi, M.; Razmi, J.; Tavakkoli-Moghaddam, R. Multiobjective invasive weed optimization for stochastic green hub location routing problem with simultaneous pick-ups and deliveries. *Econ. Comput. Econ. Cybern. Stud.* 2013, 47, 247–266.
- Govindan, K.; Jafarian, A.; Khodaverdi, R.; Devika, K. Two-echelon multiple-vehicle location-routing problem with time windows for optimization of sustainable supply chain network of perishable food. *Int. J. Prod. Econ.* 2014, *152*, 9–28, doi:10.1016/j.ijpe.2013.12.028.
- Nakhjirkan, S.; Rafiei, F.M. An integrated multi-echelon supply chain network design considering stochastic demand: A genetic algorithm-based solution. *Promet Traffic Transp.* 2017, 29, 391–400, doi:10.7307/ptt.v29i4.2193.
- Validi, S. Low-Carbon Multiobjective Location-Routing in Supply Chain Network Design. Ph.D. Thesis, Dublin City University Business School, Berlin, Germany, 2014.
- Validi, S.; Bhattacharya, A.; Byrne, P.J. Integrated low-carbon distribution system for the demand side of a product distribution supply chain: A DoE-guided MOPSO optimizer-based solution approach. *Int. J. Prod. Res.* 2014, 52, 3074–3096, doi:10.1080/00207543.2013.864054.
- Validi, S.; Bhattacharya, A.; Byrne, P.J. A case analysis of a sustainable food supply chain distribution system-a multiobjective approach. *Int. J. Prod. Econ.* 2014, *152*, 71–87, doi:10.1016/j.ijpe.2014.02.003.
- Faraji, F.; Afshar-Nadjafi, B. A bi-objective green location-routing model and solving problem using a hybrid metaheuristic algorithm. *Int. J. Logist. Syst. Manag.* 2018, 30, 366–385, doi:10.1504/IJLSM.2018.092615.
- Tang, J.; Ji, S.; Jiang, L. The design of a sustainable location-routing-inventory model considering consumer environmental behavior. *Sustainability* 2016, *8*, 211, doi:10.3390/su8030211.

- Qazvini, Z.E.; Amalnick, M.S.; Mina, H. A green multi-depot location routing model with split-delivery and time window. *Int. J. Manag. Concepts Philos.* 2016, *9*, 271–282, doi:10.1504/IJMCP.2016.079837.
- 49. Rabbani, M.; Davoudkhani, M.; Farrokhi-Asi, H. A new multiobjective green location routing problem with heterogeneous fleet of vehicles and fuel constraint. *Int. J. Strateg. Decis. Sci.* 2017, *8*, 99–119, doi:10.4018/IJSDS.2017070105.
- Toro, E.M.; France, J.F.; Echeverri, M.G.; Guimarães, F. A multiobjective model for the green capacitated location-routing problem considering environmental impact. *Comput. Ind. Eng.* 2017, 110, 114–125, doi:10.1016/j.cie.2017.05.013.
- 51. Wang, X.; Li, X. Carbon reduction in the location routing problem with heterogeneous fleet, simultaneous pickup-delivery and time windows. In Proceedings of the 21st International Conference on Knowledge-Based and Intelligent Information and Engineering Systems (KES), Marseille, France, 6–8 September 2017; ZanniMerk, C., Frydman, C., Toro, C., Hicks, Y., Howlett, R.J., Jain, L.C., Eds.; Elsevier B.V.: Amsterdam, The Netherlands, 2017; Volume 112; pp. 1131–1140. doi:10.1016/j.procs.2017.08.147.
- Qian, Z.; Zhao, Y.; Wang, S.; Leng, L.; Wang, W. A hyper heuristic algorithm for low carbon location routing problem. In Proceedings of the Advances in Neural Networks-ISNN 2018, 15th International Symposium on NeuralNetworks, Minsk, Belarus, 25–28 June 2018. doi:10.1007/978-3-319-92537-0\_21.
- Ferreira, T.N.; Jackson, A.; Lima, P.; Strickler, A.; Kuk, J.N.; Vergilio, S.R.; Pozo, A. Hyper-heuristic-based product selection for software product line testing. *IEEE Comput. Intell. Mag.* 2017, 12, 34–45, doi:10.1109/CI.017.2670461.
- Strickler, A.; Lima, J.A.P.; Vergilio, S.R.; Pozo, A.T.R. Deriving products for variability test of feature models with a hyper-heuristic approach. *Appl. Soft Comput.* 2016, 49, 1232–1242, doi:10.1016/j.asoc.2016.07. 059.
- Walker, J.D.; Ocha, G.; Gendreau, M.; Burke, E.K. Vehicle routing and adaptive iterated local search within the HyFlex hyper-heuristic framework. In Proceedings of the Learning and Intelligent Optimization 6th International Conference, Paris, France, 16–20 January 2012. doi:10.1007/978-3-642-34413-8\_19.
- Denzinger, J.; Fuchs, M.; Fuchs, M. High performance ATP systems by combining several AI methods. In Proceedings of the International Joint Conference on Artificial Intelligence, Nagoya, Japan, 23–29 August 1997.
- 57. Cowling, P.; Kendall, G.; Soubeiga, E. A hyper-heuristic approach to scheduling a sales summit. In Proceedings of the International Conference on the Practice and Theory of Automated Timetabling, Konstanz, Germany, 16–18 August 2000; Burke, E., Erben, W., Eds.; Springer: Berlin, Germany, 2000. doi:10.1007/3-540-44629-X\_11.
- 58. Burke, E.K.; Gendreau, M.; Hyde, M.; Kendall, G.; Ochoa, G.; Qu, R. Hyper-heuristics: A survey of the state of the art. J. Oper. Res. Soc. 2013, 64, 1695–1724, doi:10.1057/jors.2013.71.
- Kumari, A.C.; Srinivas, K. Hyper-heuristic approach for multiobjective software module clustering. J. Syst. Softw. 2016, 117, 384–401, doi:10.1016/j.jss.2016.04.007.
- 60. Li, W.; Ozcan, E.; John, R. Multiobjective evolutionary algorithms and hyper-heuristics for wind farm layout optimization. *Renew. Energy* 2017, *105*, 473–482, doi:10.1016/j.renene.2016.12.022.
- Maashi, M.; Kendall, G.; Ozcan, E. Choice function based hyper-heuristics for multiobjective optimization. *Appl. Soft Comput.* 2015, 28, 312–326, doi:10.1016/j.asoc.2014.12.012.
- Chakhlevitch, K.; Cowling, P. Hyperheuristics: Recent developments. In Adaptive and Multilevel Metaheuristics; Springer: Berlin, Germany, 2008; pp. 3–29. doi:10.1007/978-3-540-79438-7\_1.
- Burke, E.; Kendall, G.; Newall, J.; Hart, E.; Ross, P.; Schulenburg, S. Hyper-heuristics: An emerging direction in modern search technology. In *Handbook of Metaheuristics*; Springer: Boston, MA, USA, 2003; pp. 457–474. doi:10.1007/0-306-48056-5\_16.
- 64. Maashi, M.; Ozcan, E.; Kendall, G. A multiobjective hyper-heuristic based on choice function. *Expert Syst. Appl.* **2014**, *41*, 4475–7793, doi:10.1016/j.eswa.2013.12.050.
- Koulinas, G.; Kotsikas, L.; Anagnostopoulos, K. A particle swarm optimization based hyper-heuristic algorithm for the classic resource constrained project scheduling problem. *Inf. Sci.* 2014, 277, 680–693, doi:10.1016/j.ins.2014.02.155.
- Kareb, D.E.; Fouquet, F.; Traon, Y.L.; Bourcier, J. Sputnik: Elitist Artificial Mutation Hyper-Heuristic for Runtime Usage of Multiobjective Evolutionary Algorithms. 2014. Available online: https://arxiv.org/abs/1402.4442v1 (accessed on 5 January 2019).
- Castro, O.R.; Pozo, A. A MOPSO based on hyper-heuristic to optimize many-objective problems. In Proceedings of the IEEE Symposium on Swarm Intelligence (SIS), Orlando, FL, USA, 9–12 December 2014; IEEE: New York, NY, USA, 2014; pp. 251–258. doi:10.1109/SIS.2014.7011803.

- Castro, O.R.; Pozo, A. Using hyper-heuristic to select leader and archiving methods for many-objective problems. In Proceedings of the 8th International Conference on Evolutionary Multi-Criterion Optimization (EMO), Guimaraes, Portugal, 29 March–1 April 2015; GasparCunha, A.; Antunes, C.H.; Coello, C.C., Eds.; Springer: Berlin, Germany, 2015; Volume 9018; pp. 109–123. doi:10.1007/978-3-319-15934-8\_8.
- Goncalves, R.A.; Kuk, J.N.; Almeida, C.P.; Venske, S.M. MOEA/D-HH: A hyper-heuristic for multiobjective problems. In *Lecture Notes in Computer Science, 8th International Conference on Evolutionary Multi-Criterion Optimization (EMO), Guimaraes, Portugal, 29 March–1 April 2015;* GasparCunha, A., Antunes, C.H., Coello, C.C., Eds.; Springer: Berlin, Germany, 2015, Volume 9018; pp. 94–108. doi:10.1007/978-3-319-15934-8\_7.
- Hitomi, N.; Selva, D. Experiments with human integration in asynchronous and sequential multi-agent frameworks for architecture optimization. In *Procedia Computer Science, Conference on Systems Engineering Research, Hoboken, NJ, USA, 17–19 March 2015*; Wade, J., Cloutier, R., Eds.; Elsevier: Amsterdam, The Netherlands, 2015; Volume 44; pp. 393–402. doi:10.1016/j.procs.2015.03.024.
- Qian, C.; Tang, K.; Zhou, Z.H. Selection hyper-heuristics can provably be helpful in evolutionary multiobjective optimization. In *Lecture Notes in Computer Science*, 14th International Conference on Parallel Problem Solving from Nature (PPSN), Edinburgh, ENGLAND, 17–21 September 2016; Handl, J., Hart, E., Lewis, P.R., LopezIbanez, M., Ochoa, G., Paechter, B., Eds.; Springer: Cham, Switzerland, 2016; Volume 9921; pp. 835–846. doi:10.1007/978-3-319-45823-6\_78.
- Freitag, M.; Hildebrandt, T. Automatic design of scheduling rules for complex manufacturing systems by multiobjective simulation-based optimization. *CIRP Ann. Manuf. Technol.* 2016, 65, 433–436, doi:10.1016/j.cirp.2016.04.066.
- 73. Guizzo, G.; Vergilio, S.R.; Pozo, A.T.R.; Fritsche, G.M. A multiobjective and evolutionary hyper-heuristic applied to the integration and test order problem. *Appl. Soft Comput.* **2017**, *56*, 331–344, doi:10.1016/j.asoc.2017.03.012.
- Hitomi, H.; Selva, D.A. classification and comparison of credit assignment strategies in multiobjective adaptive operator selection. *IEEE Trans. Evol. Comput.* 2017, 21, 294–314, doi:10.1109/TEVC.2016.2602348.
- 75. Xu, C.; Liu, Y.; Li, P.; Yang, Y. Unified multiobjective mapping for network-on-chip using genetic-based hyper-heuristic algorithms. *IET Comput. Digit. Tech.* **2017**, *12*, 158–166, doi:10.1049/iet-cdt.2017.0156.
- Yao, Y.; Peng, Z.; Xiao, B. Parallel hyper-heuristic algorithm for multiobjective route planning in a smart city. *IEEE Trans. Veh. Technol.* 2018, 67, 10307–10318, doi:10.1109/TVT.2018.2868942.
- Almeida, C.; Goncalves, R.; Venske, S.; Luders, R.; Delgado, M. Multi-armed bandit based hyper-heuristics for the permutation flow shop problem. In Proceedings of the 7th Brazilian Conference on Intelligent Systems (BRACIS), Sao Paulo, Brazil, 22–25 October 2018; IEEE: New York, NY, USA, 2018; pp. 139–144. doi:10.1109/BRACIS.2018.00032.
- Gomez, J.; Terashima-Marin, H. Evolutionary hyper-heuristics for tackling bi-objective 2D bin packing problems. *Genet. Program. Evol. Mach.* 2018, 19, 151–181, doi:10.1007/s10710-017-9301-4.
- 79. Castro, O.R.; Fritsche, G.M.; Pozo, A. Evaluating selection methods on hyper-heuristic multiobjective particle swarm optimization. *J. Heuristics* **2018**, *24*, 581–616, doi:10.1007/s10732-018-9369-x.
- Zhang, Y.; Harman, M.; Ochoa, G.; Rule, G.; Brinkkemper, S. An empirical study of meta-and hyperheuristic search for multiobjective release planning. *ACM Trans. Softw. Eng. Methodol.* 2018, 27, 3, doi:10.1145/3196831.
- Zhou, Y.; Yang, J.; Zheng, L. Hyper-heuristic coevolution of machine assignment and job sequencing rules for multiobjective dynamic flexible job shop scheduling. *IEEE Access* 2019, 7, 68–88, doi:10.1109/ACCESS.2018.2883802.
- Chand, S.; Singh, H.; Ray, T. Evolving heuristics for the resource constrained project scheduling problem with dynamic resource disruptions. *Swarm Evol. Comput.* 2019, 44, 897–912, doi:10.1016/j.swevo.2018.09.007.
- 83. Li, W.; Ozcan, E.; John, R. A learning automata-based multiobjective hyper-heuristic. *IEEE Trans. Evol. Comput.* **2019**, *23*, 59–72, doi:10.1109/TEVC.2017.2785346.
- 84. Krause, E.F. Taxicab Geometry: An Adventure in Non-Euclidean Geometry; Dover Publisher: New York, NY, USA, 2012.
- Karaoglan, I.; Altiparmak, F.; Kara, I.; Dengiz, B. A branch and cut algorithm for the location-routing problem with simultaneous pickup and delivery. *Eur. J. Oper. Res.* 2011, 211, 318–332, doi:10.1016/j.ejor.2011.01.003.
- 86. Barreto, S.; Ferreira, C.; Paixao, J.; Santos, B.S. Using clustering analysis in a capacitated location-routing problem. *Eur. J. Oper. Res.* **2007**, *179*, 968–977, doi:10.1016/j.ejor.2006.06.074.
- Deb, K.; Pratap, A.; Agarwal, S.; Meyarivan, T. A fast and elitist multiobjective genetic algorithm: NSGA-II. *IEEE Trans. Evol. Comput.* 2002, *6*, 182–197, doi:10.1109/4235.996017.

- Zitzler, E.; Laumanns, M.; Thiele, L. SPEA2: Improving the strength Pareto evolutionary algorithm. In Proceedings of the Evolutionary Methods for Design, Optimization and Control with Applications to Industrial Problems, Athens, Greece, 19–21 September 2001.
- 89. Li, M.; Yang, S.; Liu, X. Bi-goal evolution for many-objective optimization problems. *Artif. Intell.* **2015**, *228*, 46–65, doi:10.1016/j.artint.2015.06.007.
- 90. Chen, B.; Zeng, W.; Lin, Y.; Zhang, D. A new local search-based multiobjective optimization algorithm. *IEEE Trans. Evol. Comput.* **2015**, *19*, 50–73, doi:10.1109/TEVC.2014.2301794.
- 91. Yang, S.; Li, M.; Liu, X.; Zheng, J. A grid-based evolutionary algorithm for many-objective optimization. *IEEE Trans. Evol. Comput.* **2013**, *5*, 721–736, doi:10.1109/TEVC.2012.2227145.
- Zitzler, E.; Kunzli, S. Indicator-based selection in multiobjective search. In Proceedings of the Parallel Problem Solving from Nature-PPSN VIII, International Conference on Parallel Problem Solving from Nature, Birmingham, UK, 13–17 September 2004; Yao, X., Ed.; Springer: Berlin, Germany, 2004; doi:10.1007/978-3-540-30217-9\_84.
- Deb, K.; Jain, H. An evolutionary many-objective optimization algorithm using reference-point based nondominated sorting approach, Part I: Solving problems with box constraints. *IEEE Trans. Evol. Comput.* 2014, 4, 577–601, doi:10.1109/TEVC.2013.2281535.
- Corne, D.W.; Jerram, N.R.; Knowles, J.D.; Oates, M.J. PESA-II: Region-based selection in evolutionary multiobjective optimization. In Proceedings of the Genetic and Evolutionary Computation Conference, New York, NY, USA, 9–13 July 2002.
- 95. Zhang, Q.; Li, H. A multiobjective evolutionary algorithm based on decomposition. *IEEE Trans. Evol. Comput.* **2007**, *11*, 712–731, doi:10.1109/TEVC.2007.892759.
- Scoring System. Available online: http://www.asap.cs.nott.ac.uk/external/chesc2011/ (accessed on 12 February 2019).
- Nadizadeh, A.; Kafash, B. Routing problem with simultaneous pickup and delivery demands. *Transp. Lett.* 2019, 1, 1–19, doi:10.1080/19427867.2016.1270798.
- Yu, H.; Solvang, W.D. A carbon-constrained stochastic optimization model with augmented multi-criteria scenario-based risk-averse solution for reverse logistics network design under uncertainty. *J. Clean. Prod.* 2017, 164, 1248–1267, doi:10.1016/j.jclepro.2017.07.066.
- Yu, H.; Solvang, W.D. Incorporating flexible capacity in the planning of a multi-product multi-echelon sustainable reverse logistics network under uncertainty. J. Clean. Prod. 2018, 198, 285–303, doi:10.1016/j.jclepro.2018.07.019.
- 100. Yu, H.; Solvang, W.D. An improved multiobjective programming with augmented ε-constraint method for hazardous waste location-routing problems. *Int. J. Environ. Res. Public Health* **2016**, *13*, 548, doi:10.3390/ijerph13060548.



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).